

OPEN ACCESS

Operations Research and Decisions

www.ord.pwr.edu.pl

OPERATIONS RESEARCH AND DECISIONS QUARTERLY

erdí rí

Solution of an uncertain EPQ model using the Neutrosophic differential equation approach

Mostafijur Rahaman¹^b Rakibul Haque²^b Soheil Salahshour³ Anna Sobczak⁴ Fariba Azizzadeh⁵ Shariful Alam¹^b Sankar Prasad Mondal^{*2}^b

¹Department of Mathematics, School of Liberal Arts and Sciences, Mohan Babu University, Tirupati–517102, Andhra Pradesh, India ²Department of Applied Mathematics, Maulana Abul Kalam Azad University of Technology, West Bengal, Haringhata, Nadia-741249, West

Bengal, India

³ Faculty of Engineering and Natural Sciences, Istanbul Okan University, Istanbul, Turkey

⁴Faculty of Economics, The Jacob of Paradies University in Gorzów, Wielkopolski, Poland

⁵Department of Management, Islamic Azad University, Piranshahr (Urmia), Iran

*Corresponding author, email address: sankar.res07@gmail.com

Abstract

Time is a very crucial factor in controlling demand patterns for certain products. In manufacturing processes, the production rate must be regulated according to the demand pattern and available stock as part of effective lot size management policies. We incorporate this fundamental idea for constructing the production rate as a function of demand and stock, which is the primary contribution of this paper. Predicting demand patterns and adjusting the production rate inherently involve vagueness. We use neutrosophic logic, an advanced mathematical tool for addressing imprecision in decision planning. Neutrosophic calculus-based analysis of uncertainty involved with the proposed model is the secondary contribution in this paper. Numerical results indicate that the proposed approach yields superior results compared to the crisp environment and traditional neutrosophic approaches for cost minimization. Furthermore, it is worth noting that Case 1 of the proposed Neutrosophic Differential Approach guarantees better results than Case 2.

Keywords: *EPQ* model with deterioration, Time impacted demand, Stock and demand dependent product process, Decision making under impreciseness, Neutrosophic ruled uncertainty, Triangular Neutrosophic numbers, Neutrosophic differential equation, Neutrosophic derivative

1. Introduction

Inventory represents idle resources. Therefore, inventory management involves planning and implementing policies to maintain the optimal stock size, maximizing gains while meeting consumer demand.

Received 8 May 2024, accepted 3 January 2025, published online 8 February 2025 ISSN 2391-6060 (Online)/© 2025 Authors

This is not yet the definitive version of the paper. This version will undergo additional copyediting, typesetting and review before it is published in its final form, but we are providing this version to give early visibility of the article.

Inventory management is relevant in both the retail and manufacturing sectors. In manufacturing, production inventory management includes both the manufacturing process and the maintenance of the produced items to optimize average costs. In this context, lot size models concerning order size and production size fulfilled the purpose. In this paper, an economic production quantity (EPQ) model is designed and analyzed using mathematical tools. Demand and production are two important issues impacting the manufacturing-supply-inventory environment. It is noted that demand may increase as time advances in a newly constituted supply enterprise. Therefore, the demand for an EPQ model can be viewed as a function of time. The second notable issue is the dependence of production rates on multiple factors. It is perceived that demand immediately impacts the production rate. Because the managerial body of a manufacturing organization must be bothered about the demand rate while taking measures on production capacity, another important issue is the influence of the present inventory level on the production rate. It is customary to reduce the production rate when the inventory level is already high in the store to avoid carrying costs. Also, the deterioration of production during the inventory inventory-carrying procedure cannot be overlooked. This concern motivates us to design the proposed EPQ model with new insights. Primarily, the model is built in a deterministic environment. However, the consideration of such a deterministic environment may restrict a mathematical model far from the real economic interactions involving uncertainty. Therefore, we find the necessity of uncertainty environments described by aptly fitted mathematical tools. The study of this paper is engaged in finding the answer to the following questions:

- (i) Time impacts demand. What will be the overall impact of time on minimizing the average cost ?
- (ii) Production rate is influenced by demand and present stock. What are the roles of these issues in cost minimization objectives?
- (iii) Deterioration is an unavoidable factor related to the inventory-carrying procedure. What is the subsequent impact of the deterioration factor on cost reduction?
- (iv) The impreciseness involved in economic production strategy cannot be ignored. What will be the mathematical tool and approach to encountering such phenomena with the mentioned objective and a meaningful perspective?

With the above-mentioned questionaries, the existing literature has been surveyed, which is detailed in the succeeding section.

The remaining text in this paper is structured in pockets as follows: Section 2 summarizes the literature survey and gaps in the existing literature. Section 3 provides the mathematical preliminaries, which help the reader understand the mathematical foundation of this paper. Section 4 describes the notations and symbols in the paper and their meanings. Also, the same section discusses the assumptions for the mathematical formulation of the proposed model. Section 5 details the proposed model in a crisp environment. Subsequently, Section 6 reconsiders the proposed model under neutrosophic uncertainty. The crispification of the neutrosophic model is described in Section 7. Section 8 is about the numerical results of the proposed model in different environments and approaches. Section 9 lists significant findings and managerial interpretations. The concluding remarks on the investigation and findings of the paper are given in Section 10.

2. Literature review

The literature survey is performed on keywords like recent literature on the EPQ model in a crisp environment, inventory models under different types of uncertainty, and recent advancements in neutrosophic logic and its applications. The subsections corresponding to each mentioned keyword are followed by subsections concerning research gaps, motivations, and the contribution of the paper.

2.1. Recent literature on the EPQ model in a crisp environment

Cárdenas-Barrón [10] proposed an EPQ model incorporating planned backorders. This system produces imperfect-quality items, all of which are reworked within the same production cycle. A deteriorating EPQ model with multiple manufacturing stages and one remanufacturing stage is introduced by Widyadana and Wee [60]. Taleizadeh et al. [57] formulated an EPQ model that incorporates random defective items, repair failures, shortages, and the presence of a single machine, which leads to constrained production capacity. A production-supply model with learning-based production cost is developed by Teng et al. [58] from the seller's perspective to determine the optimal lot size and trade credit tenure simultaneously. Hsu and Hsu [22] presented a defective production-based EPQ model that allows for complete backlog shortages. A learning-based EPQ model is investigated by Khan et al. [25], where learning is imposed on production rate and demand is dependent on variable lead time stochastically. Cunha et al. [13] proposed an EPQ model considering partial backlogged shortages and defective production batches. They demonstrate that selling imperfect items promptly is preferable since the reduction in holding costs leads to an overall cost decrease. Taleizadeh et al. [56] proposed an EPQ model considering sustainability with three different scenarios of shortage, namely, partial backordered shortage, lost sale, and full backorder shortage. They found that a partial backlogged shortage case is the best and most realistic model. A defective production-based EPQ model is presented by Keshavarzfard et al. [24], where the production rate is dependent on the demand pattern. Marchi et al. [30] examined a production inventory model by applying the learning-by-doing approach to the production rate and maintaining the quality of items, which leads to the efficiency of energy and reliability of the system. They demonstrated the interconnectedness between production learning and energy efficiency and how this relationship influences the optimal quantity of lot size. Taleizadeh et al. [55] introduced a reworked-based EPQ system with price-dependent demand-taking pricing, producing quantity, and back-ordered quantity as decision variables. Nobil et al. [38] studied an EPQ model for a defective production system with remanufacturing and shortages under a 100% inspection process. Recently, Rahaman et al. [42] designed an imperfect production-based EPQ model where the demand rate is influenced by the frequency of promotion, the greenness of the item, and the selling price, and the rate of producing defective items is linearly dependent on time. Haque et al. [20] presented a sustainable production inventory model with price- and greenness-dependent demand and demand-dependent remanufacturing rates. In this study, a deteriorating EPQ model is analyzed, where production rate is influenced by demand and on-hand stock level, and demand is influenced by time.

2.2. Inventory models under different types of uncertainty

After fuzzy set theory was introduced, Park [40] was the first to incorporate fuzzy concepts into an Economic Order Quantity (EOQ) model. In the inventory control problem, several research articles are published incorporating fuzzy uncertainty, such as work by Hojati [21], Bag et al. [8], De and Sana [14], Sadeghi et al. [45], Sarkar and Mahapatra [48], Majumder et al. [28], and Mahata et al. [26]. Debnath et al. [17] investigated a sustainable EPQ model in a type-2 fuzzy uncertain environment where the demand pattern is dependent on inventory level and selling price and the production rate is dependent on the demand rate. De and Sana [15, 16] used the notion of intuitionistic fuzzy in controlling strategy. Garai et al. [18] used intuitionistic fuzzy numbers to measure the uncertainty in an inventory system with stock-dependent demand. A good number of studies have been published on inventory models using intuitionistic fuzzy, such as Ali et al. [5], Sahoo et al. [46], Supakar et al. [54], and Giri et al. [19]. Momena et al. [34] discussed an EOQ model with price-dependent demand under all unit price discount policies in a densely fuzzy environment. A sustainable production and rework model is developed in a dense-lock fuzzy environment by Karmakar et al. [23]. Rahaman et al. [44] studied an EPQ model in a lock fuzzy environment where the demand rate is dependent on price, inventory level, and deterioration of the item using a preservation facility. Maiti [27] studied an EPQ model with imperfect production and demand-dependent production rates in an uncertain arena by taking the price of the produced item as a fuzzy cloud number and using PSO to solve the problem. Barman et al. [9] investigated a deteriorating EPQ model with a partial backlogged shortage and time-dependent demand in the cloud fuzzy phenomenon. Manna et al. [29] studied a production inventory model with greenlevel-dependent demand, considering carbon emissions during production in an uncertain environment. Rahaman et al. [43] interpreted an EOQ model with price and stock-dependent demand in a type-2 interval uncertain environment.

2.3. Recent advancements in neutrosophic logic and its application

The introduction of the neutrosophic philosophy triggers enthusiasm among the researcher communities around the globe. Classification of different types of neutrosophic numbers, namely single-valued (Wang et al. [59]), triangular (Chakraborty et al. [12]), trapezoidal (Ye [61]), pentagonal (Chakraborty [11]), and complex (Ali and Smarandache [6]), neutrosophic numbers, has been done subsequently, along with their possible application in the fields of engineering and management. The intuition and sense of neutrosophic philosophy had been further enriched by several worthy findings, such as the neutrosophic triplet group by Smarandache and Ali [50], the neutrosophic vector space by Agboola and Akinleye [4], and the neutrosophic topological space by Salama and Alblowi [47]. The fuzzy derivative of a fuzzy valued function was first introduced by Puri and Ralescu [41]. Several researchers [1–3, 7] applied this concept to their research. Smarandache [49] introduced the concept of the neutrosophic derivative within the neutrosophic realm as an expansion of the fuzzy derivative. Son et al. [51] gave a novel definition of the neutrosophic derivative, namely, the granular derivative providing the necessary and sufficient state for the granular derivative of a neutrosophic-valued function. Sumathi and Priya [53] and Sumathi and Sweety [52] discussed an NDE based on the parametric representation of the neutrosophic number. However, Moi et al. [31] introduced a new type of neutrosophic derivative, which is known as a gener-

alized neutrosophic derivative and used this definition to discuss a second-order neutrosophic boundary value problem in [32]. Several researchers used neutrosophic logic in various fields, including control theory and decision-making. Mullai and Surya [36, 37] used triangular neutrosophic numbers in the lotsizing model. Pal and Chakraborty [39] used the area removal technique method to optimize a triangular neutrosophic-based economic order quantity model. Mondal et al. [35] studied a lot-sizing model for deteriorating seasonal products under partial backordering and time-dependent demand. Up to date, the neutrosophic differential equation (NDE) has little use in solving uncertain inventory control problems. Momena et al. [33] used NDE to discuss an uncertain EOQ model where the market demand pattern is dependent on the stock level, price, and warranty time of the item. In this article, the NDE approach is used to solve a production quantity model where demand-related parameters and deterioration rates are taken as neutrosophic numbers.

2.4. Research gaps and motivations

The following Table 1 shows the comparison between the literature in the related keywords and the proposed model:

| References | Model types | Stock-dependent | Demand-dependent | Time-dependent | Deterioration | Model environment | Solution Methodology |
|---------------------------|-------------|-----------------|------------------|----------------|---------------|-------------------|--|
| | | production rate | production rate | demand | | | |
| Khan et al. [25] | EPQ | × | × | \checkmark | × | Crisp | Differential equation |
| Keshavarzfard et al. [24] | EPQ | × | \checkmark | | X | Crisp | Differential equation |
| Haque et al. [20] | EPQ | × | | × | x X | Crisp | Differential equation |
| Majumder et al. [28] | EPQ | × | × | \checkmark | \checkmark | Fuzzy | Fuzzy differential equation |
| Debnath et al. [17] | EPQ | × | \checkmark | × | × | Type-2 fuzzy | Fuzzy differential equation |
| Rahaman et al. [44] | EPQ | \checkmark | × | × | \checkmark | Lock fuzzy | Parametric representation of lock fuzzy number |
| Maiti [27] | EPQ | × | \checkmark | × | × | Cloudy fuzzy | Parametric representation of the cloudy fuzzy number |
| Barman et al. [9] | EPQ | × | × | \checkmark | \checkmark | Cloudy fuzzy | Parametric representation of the cloudy fuzzy number |
| Mondal et al. [35] | EOQ | - | - 🔺 | \checkmark | \checkmark | Neutrosophic | Parametric representation of the neutrosophic number |
| Momena et al. [33] | EOQ | - | | × | × | Neutrosophic | Neutrosophic differential equation |
| This article | EPQ | \checkmark | \checkmark | \checkmark | \checkmark | Neutrosophic | Neutrosophic differential equation |

Table 1. Comparison of the contributions among the literature related to the present paper

From the detailed research survey on the above-mentioned keywords, we have found the following research gaps which are targeted to be overcome in the present article:

- (i) Demand with time dependency was discussed in many existing models. However, the demanddependent production rate is rarely considered in the existing literature. In this paper, the production rate is assumed to be demand- and stock-dependent.
- (ii) Many economic scenarios under impreciseness were discussed using mathematical tools like different fuzzy and interval numbers. Among these studies, only a few adapted uncertain differential equation approaches driven by interval and fuzzy-valued calculus. This lacuna motivates us to consider the proposed EPQ model under an uncertain decision environment and to describe it using uncertain differential equations.
- (iii) The philosophy of neutrosophy extends the idea of uncertainty incurred by fuzzy and intuitionistic fuzzy logic. A fuzzy set comprises the notion of membership of elements generalizing the classical set. An intuitionistic fuzzy has the additional feature of a non-membership grade. Furthermore, a neutrosophic set contributes a more structured and generalized mathematical sense of uncertainty with membership, non-membership, and hesitancy grades. Thus, we feel the necessity of discussing the proposed EPQ model under uncertainty using neutrosophic sets and numbers.

(iv) Surveying the applications of the neutrosophic numbers in the inventory models, we find some EOQ and EPQ models in the existing literature. However, in those studies, the de-neutrosophication technique is used very before the optimization. In other words, the dynamics of the neutrosophic-based system in those studies are described in terms of crisp calculus. In this paper, the neutrosophic differential equation is used to describe an inventory control model.

2.5. Novelty

The current article contributes to some novel perspectives on theoretical advancement. They are:

- (i) Time has an obvious impact on the demand rate. In many newly organized supply bodies, the enthusiasm increases linearly with time. This concern was addressed in much literature. Also, present stock controls the production rate of a manufacturing body, and this issue has been investigated by many authors. However, we cannot find a single piece of literature where the impact of time on the production rate through demand patterns and the stock has been traced. The proposed model is distinguished in this context.
- (ii) The proposed crisp model is analyzed, and the optimality criteria have been found in the cost minimization objective. Then, we introduced uncertainty using the neutrosophic number, which carries a more generalized sense of uncertainty compared to fuzzy and intuitionistic fuzzy numbers. The theory of neutrosophic calculus and differential equations has been employed to analyze the uncertain version of the proposed EPQ model.
- (iii) In this current article, a novel de-neutrosophication formula has been introduced.
- (iv) Moreover, the new de-neutrosophication formula, along with the neutrosophic differential equation approach, provides better results compared to the crisp and old neutrosophic optimization techniques regarding the cost minimization objective.

3. Mathematical preliminaries

Definition 1. [12] A fuzzy set is denoted by the pair $(x, \mu(x))$, where x is an element in the universal set X, and $\mu(x)$ represents the degree to which x belongs to X, with $\mu(x)$ falling within the range [0, 1].

Definition 2. [12] An intuitionistic fuzzy set is represented by the ordered triplet $(x, \mu(x), \nu(x))$, where x is an element in the universal set X, $\mu(x)$ and $\nu(x)$ represents the degree of belongingness and non-belongingness of x in X, respectively, and both $\mu(x)$ and $\nu(x)$ are within the range [0, 1] satisfying the condition $0 \le \mu(x) + \nu(x) \le 1$.

Definition 3. [12] A neutrosophic set is represented by the ordered triplet (x, T(x), I(x), F(x)), where x is an element in the universal set X, T(x), I(x) and F(x) respectively signify the degrees of truthiness, indeterminacy, and falsity of x in X. Each of T(x), I(x) and F(x) lies in the range [0, 1], fulfilling the condition $0 \le T(x) + I(x) + F(x) \le 3$.

Definition 4. [12] A single-valued triangular neutrosophic (SVTN) number of Type 1, denoted as $\tilde{G}_{TN} = (d_1, d_2, d_3; s_1, s_2, s_3; z_1, z_2, z_3)$ is a specific type of neutrosophic set on the real numbers R. It is characterized by its truth, indeterminacy, and falsity membership functions, which are defined as follows:

$$T_{\tilde{G}_{TN}}(x) = \begin{cases} \frac{x-d_1}{d_2-d_1} & \text{when } d_1 \leq x \leq d_2 \\ 1 & \text{when } x = d_2 \\ \frac{d_3-x}{d_3-d_2} & \text{when } d_2 \leq x \leq d_3 \\ 0 & \text{otherwiae} \end{cases}$$

$$I_{\tilde{G}_{TN}}(x) = \begin{cases} \frac{s_1-x}{s_2-s_1} & \text{when } s_1 \leq x \leq s_2 \\ 0 & \text{when } x = s_2 \\ \frac{x-s_2}{s_3-s_2} & \text{when } s_2 \leq x \leq s_3 \\ 1 & \text{otherwiae} \end{cases}$$

$$F_{\tilde{G}_{TN}}(x) = \begin{cases} \frac{z_1-x}{s_2-z_1} & \text{when } z_1 \leq x \leq z_2 \\ 0 & \text{when } x = z_2 \\ \frac{x-z_2}{z_3-z_2} & \text{when } z_2 \leq x \leq z_3 \\ 1 & \text{otherwiae} \end{cases}$$

and $0 \le T_{\tilde{G}_{TN}}(x) + I_{\tilde{G}_{TN}}(x) + F_{\tilde{G}_{TN}}(x) \le 3.$

Definition 5. [12] The (α, β, γ) -cut of a neutrosophic set $\tilde{G} = (x, T_{\tilde{G}}(x), I_{\tilde{G}}(x), F_{\tilde{G}}(x))$ over X is indicated by $[\tilde{G}]_{\alpha,\beta,\gamma}$ and is defined by $[\tilde{G}]_{\alpha,\beta,\gamma} = \{\langle x, T_{\tilde{G}}(x), I_{\tilde{G}}(x), F_{\tilde{G}}(x) \rangle : T_{\tilde{G}} \ge \alpha, I_{\tilde{G}}(x) \le \beta, F_{\tilde{G}} \le \gamma\}$. It is also known as a parametric representation or parametric form of the neutrosophic set.

Note: The parametric representation of an SVTN number $\tilde{G}_{TN}N = (d_1, d_2, d_3; s_1, s_2, s_3; z_1, z_2, z_3)$ includes six components. These six components are written as $\langle [G_1(\alpha), G_2(\alpha)], [G'_1(\beta), G'_2(\beta)], [G''_1(\gamma), G''_2(\gamma)] \rangle$, where $G_1(\alpha) = d_1 + \alpha(d_2 - d_1), G_2(\alpha) = d_3 - \alpha(d_3 - d_2), G'_1(\beta) = s_2 - \beta(s_2 - s_1), G''_2(\beta) = s_2 + \beta(s_3 - s_2), G''_1(\gamma) = z_2 - \gamma(z_2 - z_1) \text{ and } G''_2(\gamma) = z_2 + \gamma(z_3 - z_2).$

Definition 6. [31] Let $\tilde{g} : I \to N$ be a neutrosophic-valued function given in the parametric representation by $\tilde{g}(t) = \langle [g_1(t;\alpha), g_2(t;\alpha)], [g'_1(t;\beta), g'_2(t;\beta)], [g''_1(t;\gamma), g''_2(t;\gamma)] \rangle, \forall t \in I$. The generalized neutrosophic derivative of $\tilde{g}(t)$ at $t = c \in I$ is written as $\tilde{g}(c) = \langle \dot{g}_T(c), \dot{g}_I(c), \dot{g}_F(c) \rangle$ in which $\dot{g}_T(c)$, $\dot{g}_I(c)$ and $\dot{g}_F(c)$ are defined as the following

1.
$$\dot{g}_T(c) = [min\{\dot{g}_1(c;\alpha), \dot{g}_2(c;\alpha)\}, max\{\dot{g}_1(c;\alpha), \dot{g}_2(c;\alpha)\}]$$

- 2. $\dot{g}_I(c) = [min\{\dot{g}'_1(c;\beta),\dot{g}'_2(c;\beta)\}, max\{\dot{g}'_1(c;\beta),\dot{g}'_2(c;\alpha)\}]$
- 3. $\dot{g}_F(c) = [min\{\dot{g}_1''(c;\gamma), \dot{g}_2''(c;\gamma)\}, max\{\dot{g}_1''(c;\gamma), \dot{g}_2''(c;\gamma)\}]$

provided $\dot{g}_1(c;\alpha)$, $\dot{g}_2(c;\alpha)$, $\dot{g}'_1(c;\beta)$, $\dot{g}'_2(c;\beta)$, $\dot{g}''_1(c;\gamma)$ and $\dot{g}''_2(c;\gamma)$ are all exists. $\tilde{\dot{g}}(c)$ is said to be a type-1 derivative if the parametric representation of $\tilde{\dot{g}}(c)$ is given by

$$\tilde{\dot{g}}(c) = \langle [\dot{g}_1(c;\alpha), \ \dot{g}_2(c;\alpha)], [\dot{g}_1'(c;\beta), \dot{g}_2'(c;\beta)], [\dot{g}_1''(c;\gamma), \ \dot{g}_2''(c;\gamma)] \rangle$$

and type-2 derivative if the parametric representation of $\tilde{\dot{g}}(c)$ is given by

$$\tilde{\dot{g}}(c) = \langle [\dot{g}_2(c;\alpha), \ \dot{g}_1(c;\alpha)], [\dot{g}_2'(c;\beta), \dot{g}_1'(c;\beta)], [\dot{g}_2''(c;\gamma), \ \dot{g}_1''(c;\gamma)] \rangle$$

4. Notations and assumptions of the proposed model

To explain the proposed model, the subsequent symbols and presumptions are employed.

4.1. Notations

All the notations related to the proposed model are described in Table 2.

| Tuble 2. Descriptions of the notations and then description units | | | | | | | | |
|---|--|---------------------|--|--|--|--|--|--|
| Notations | Explanation | Units | | | | | | |
| a | The constant part of the demand pattern | Constant | | | | | | |
| b | Coefficient of time in the demand pattern | Constant | | | | | | |
| m | A constant part of the production rate | Constant | | | | | | |
| n | Coefficient of inventory level in the production rate | Constant | | | | | | |
| l | Coefficient of the demand pattern in the production rate | Constant | | | | | | |
| θ | Rate of deterioration | Constant | | | | | | |
| c_h | Holding cost per unit item per unit time | \$/item/unit time | | | | | | |
| c_p | Production cost per unit item per unit time | \$/item/unit time | | | | | | |
| C_0 | Ordering cost | \$/cycle | | | | | | |
| K | Production rate | Unit item/unit time | | | | | | |
| t_1 | Production time (decision variable) | Unit time | | | | | | |
| T | Total cycle time (decision variable) | Unit time | | | | | | |
| ϕ | Total average profit (objective function) | \$ | | | | | | |
| | | | | | | | | |

Table 2. Descriptions of the notations and their description units

K.

4.2. Assumptions

We consider the phenomenon of a newly launched production plant. There are ambiguities about the demand pattern, and the demand for the produced items increases gradually as time passes due to the recognition of the newly built production plan among the consumer communities. The production of items is dependent on the stock of items already produced and on the demand pattern. Also, the deterioration of items in stock is taken into consideration. Mathematically, the proposed model is developed based on the assumptions listed below:

- (a) The demand rate of the produced item is dependent on time, i.e., over time, the rate of demand is steadily rising as a linear function of time. D(t) = a + bt, where a, b are positive constants.
- (b) The production rate is dependent on the hand stock level as well as on the demand rate. i.e., K = m nq(t) + lD(t) = (m + al) nq(t) + lbt, where m, n, l are positive constants.
- (c) The products in stock have deteriorated at a constant rate $\theta(0 \le \theta \le 1)$ throughout the whole lotsizing cycle.
- (d) The lead time is zero.
- (e) The rate of replenishment is infinite, yet the lot size is finite.

- (f) No shortage is considered in the whole lot-sizing cycle.
- (g) The time horizon is finite.

5. Formulation of the proposed crisp EPQ model

The production inventory model is started at t = 0 with stock- and demand-dependent production rates. The production phase is stopped at $t = t_1$ making the highest possible stocks of the whole productionsupply cycle. The lot-sizing cycle has faced decay in the inventory level due to deterioration at a constant rate and a time-dependent demand rate. The whole lot-sizing cycle is terminated at t = T (see Figure 1). The mathematical correspondence of the mentioned set-up can be represented by the set of subsequent two differential equations.



Figure 1. Graphical representation of the proposed EOQ mode

The governing differential equation of the production time $(0 \le t \le t_1)$ along with the boundary conditions is given by

$$\frac{dq(t)}{dt} = \{(m+al) - nq(t) + lbt\} - \{a+bt\} - \theta q(t)$$

with q(0) = 0, $q(t_1) = Q$. That means,

$$\frac{dq(t)}{dt} = m - (1-l)a - (1-l)bt - (\theta+n)q(t)$$
(1)

and the governing differential equation of the non-production time $(t_1 \le t \le T)$ of the lot cycle is given by

$$\frac{dq(t)}{dt} = -\{a+bt\} - \theta q(t) \tag{2}$$

with q(T) = 0.

Solving the systems represented by equations (1) and (2), the following results are obtained: the stock level in the productive time interval ($0 \le t \le t_1$) is obtained as

$$q(t) = \frac{k_1 k_3 + k_2}{k_1^2} \{1 - e^{-k_1 t}\} - \frac{k_2}{k_1} t$$
(3)

where $k_1 = \theta + n$, $k_2 = b(1 - l)$ and $k_3 = m - a(1 - l)$. The stock level in the non-productive time interval $(t_1 \le t \le T)$ is given by

$$q(t) = \frac{b - \theta(a + bt)}{\theta^2} - \frac{b - \theta(a + bT)}{\theta^2} e^{\theta(T - t)}$$

$$\tag{4}$$

The highest level of stock at $t = t_1$ is obtained as

$$Q = \frac{k_1 k_3 + k_2}{k_1^2} \{1 - e^{-k_1 t_1}\} - \frac{k_2}{k_1} t_1$$
(5)

 \mathbf{V}

Using the continuity of the stock function at $t = t_1$ from equations (3) and (4), the following constraint is obtained

$$\frac{b-\theta(a+bt_1)}{\theta^2} - \frac{b-\theta(a+bT)}{\theta^2}e^{\theta(T-t_1)} = \frac{k_1k_3+k_2}{k_1^2}\{1-e^{-k_1t_1}\} - \frac{k_2}{k_1}t_1$$
(6)

Several relevant costs associated with the model are formulated as follows:

$$\begin{aligned} HC &= c_h \left[\int_0^{t_1} q(t) dt + \int_{t_1}^T q(t) dt \right] \\ &= c_h \left[\frac{k_1 k_3 + k_2}{k_1^3} \{ e^{-k_1 t_1} + t_1 k_1 - 1 \} - \frac{k_2 t_1^2}{2k_1} + \frac{b}{2\theta} (t_1^2 - T^2) + \frac{b - \theta(a + bT)}{\theta^3} \{ 1 - e^{\theta(T - t_1)} \} \right] \end{aligned}$$

Production Cost (PC): the production cost is given by

$$PC = c_p \left[\int_0^{t_1} \{(m+al) - nq(t) + lbt\} dt \right]$$

= $c_p \left[(m+al)t_1 + \frac{lbt_1^2}{2} - n \left\{ \frac{k_1k_3 + k_2}{k_1^3} \{e^{-k_1t_1} + t_1k_1 - 1\} - \frac{k_2t_1^2}{2k_1} \right\} \right]$

Setup Cost (SC): the setup cost is taken as constant C_0 throughout the inventory cycle. Therefore, the total average cost of the production system during the entire circle is given by

$$\Phi(t_1, T) = \frac{HC + PC + C_0}{T}$$

$$\begin{split} \Phi(t_1,T) &= \frac{1}{T} \Biggl[c_h \Biggl[\frac{k_1 k_3 + k_2}{k_1^3} \{ e^{-k_1 t_1} + t_1 k_1 - 1 \} - \frac{k_2 t_1^2}{2k_1} + \frac{b}{2\theta} (t_1^2 - T^2) + \frac{b - \theta(a + bT)}{\theta^3} \\ &\times \{ 1 - e^{\theta(T - t_1)} \} \Biggr] + c_p \Biggl[(m + al) t_1 + \frac{lb t_1^2}{2} - n \Biggl\{ \frac{k_1 k_3 + k_2}{k_1^3} \{ e^{-k_1 t_1} + t_1 k_1 - 1 \} \\ &- \frac{k_2 t_1^2}{2k_1} \Biggr\} \Biggr] + C_0 \Biggr] \end{split}$$

Therefore, the cost-minimization problem will be

$$\begin{cases}
Minimize \quad \Phi(t_1, T) = \frac{HC + PC + C_0}{T} \\
Subject \ to \quad Q = \frac{k_1 k_3 + k_2}{k_1^2} \{1 - e^{-k_1 t_1}\} - \frac{k_2}{k_1} t_1 \\
\quad \frac{b - \theta(a + bt_1)}{\theta^2} - \frac{b - \theta(a + bT)}{\theta^2} e^{\theta(T - t_1)} = \frac{k_1 k_3 + k_2}{k_1^2} \{1 - e^{-k_1 t_1}\} - \frac{k_2}{k_1} t_1 \\
\quad 0 \le t_1 \le T
\end{cases}$$
(7)

Here, the goal is to investigate the convexity of the average cost function $\Phi(t_1, T)$ concerning the decision variables t_1 and T.

Theorem 1. The average cost function $\Phi(t_1, T)$ is strictly pseudo-convex in t_1 and T hence, $\Phi(t_1, T)$ attains the minimum value at the point (t_1^*, T^*) , provided

$$\frac{(c_h - nc_p)(k_1k_3 + k_2)}{k_1}e^{-k_1t_1} + \frac{c_hb}{\theta} + c_plb > (c_h - nc_p)\frac{k_2}{k_1} + \frac{c_hb}{\theta}e^{\theta(T-t_1)}$$
(8)

Proof. For convenience, let us take the average cost function $\Phi(t_1, T)$ as

$$\Phi(t_1, T) = \frac{\Phi_1(t_1, T)}{\Phi_2(t_1, T)}$$

 $\Phi(t_1,T) = \frac{\Phi_1(t_1,T)}{\Phi_2(t_1,T)}$ where $\Phi_1(t_1,T) = c_h \Big[\frac{k_1 k_3 + k_2}{k_1^3} \{ e^{-k_1 t_1} + t_1 k_1 - 1 \} - \frac{k_2 t_1^2}{2k_1} + \frac{b}{2\theta} (t_1^2 - T^2) + \frac{b - \theta(a + bT)}{\theta^3} \{ 1 - e^{\theta(T - t_1)} \} \Big] + c_p \Big[(m + al) t_1 + \frac{lbt_1^2}{2} - n \Big\{ \frac{k_1 k_3 + k_2}{k_1^3} \{ e^{-k_1 t_1} + t_1 k_1 - 1 \} - \frac{k_2 t_1^2}{2k_1} \Big\} \Big] + C_0 \text{ and } \Phi_2(t_1,T) = T.$ Now, $\Phi_1(t_1,T)$ can be written as

$$\Phi_{1}(t_{1},T) = C_{0} + \frac{A}{k_{1}^{3}}(e^{-k_{1}t_{1}} + t_{1}k_{1} - 1) + c_{p}(m+al)t_{1} + \frac{B}{2}t_{1}^{2} - \frac{C}{2}t_{1}^{2} - \frac{c_{h}b}{2\theta}T^{2} + \frac{c_{h}}{\theta^{3}}\{b - \theta(a+bT)\}\{1 - e^{(\theta(T-t_{1}))}\}$$

where $A = (c_h - nc_p)(k_1k_3 + k_2), B = \frac{c_h b}{\theta} + c_p lb$ and $C = (c_h - nc_p)\frac{k_2}{k_1}$. To construct the Hessian matrix for $\Phi_1(t_1, T)$, all the first and second-order partial derivatives of $\Phi_1(t_1, T)$ concerning t_1 and T are calculated as

$$\frac{\partial \Phi_1(t_1, T)}{\partial t_1} = \frac{A}{k_1^2} (1 - e^{-k_1 t_1}) + c_p(m + al) + Bt_1 - Ct_1 + \frac{c_h}{\theta^2} \{b - \theta(a + bT)\} e^{\theta(T - t_1)}$$

$$\frac{\partial \Phi_1(t_1, T)}{\partial T} = -\frac{c_h b}{\theta} T - \frac{c_h}{\theta^2} \{b - \theta(a + bT)e^{\theta(T - t_1)}\}$$

$$\frac{\partial^2 \Phi_1(t_1, T)}{\partial t_1^2} = \frac{A}{k_1} e^{-k_1 t_1} + B - C - \frac{c_h}{\theta} \{b - \theta(a + bT)\}e^{\theta(T - t_1)}$$

$$\frac{\partial^2 \Phi_1(t_1, T)}{\partial t_1 \partial T} = -c_h(a + bT)e^{\theta(T - t_1)} = \frac{\partial^2 \Phi_1(t_1, T)}{\partial T \partial t_1}$$

$$\frac{\partial^2 \Phi_1(t_1, T)}{\partial t_1^2} = c_h(a + bT)e^{\theta(T - t_1)}$$

Therefore, the Hessian matrix corresponding to $\Phi_1(t_1, T)$ can be expressed as

$$H_{ii} = \begin{bmatrix} \frac{\partial^2 \Phi_1(t_1,T)}{\partial t_1^2} & \frac{\partial^2 \Phi_1(t_1,T)}{\partial t_1 \partial T} \\ \frac{\partial^2 \Phi_1(t_1,T)}{\partial T \partial t_1} & \frac{\partial^2 \Phi_1(t_1,T)}{\partial T^2} \end{bmatrix}$$

The first principal minor is

$$|H_{11}| = \frac{A}{k_1}e^{-k_1t_1} + B - C - \frac{c_h}{\theta}\{b - \theta(a + bT)\}e^{\theta(T - t_1)}$$

= $c_h(a + bT)e^{\theta(T - t_1)} + \frac{A}{k_1}e^{-k_1t_1} + B - C - \frac{c_hb}{\theta}e^{\theta(T - t_1)}$

Clearly, $k_1 = \theta + n > 0$. As 0 < l < 1, $k_2 = b(1 - l) > 0$. Again, the fixed part of the production rate function is greater than the fixed part of the market demand; therefore, $k_3 = (m - a) + al > 0$. Also, it is assumed that $c_h > nc_p$, therefore, A, B and C are all positive. Hence $|H_{11}| > 0$ if

$$\frac{A}{k_1}e^{-k_1t_1} + B > C + \frac{c_hb}{\theta}e^{\theta(T-t_1)}$$

The second principal minor is

$$|H_{22}| = \frac{\partial^2 \Phi_1(t_1, T)}{\partial t_1^2} \frac{\partial^2 \Phi_1(t_1, T)}{\partial T^2} - \frac{\partial^2 \Phi_1(t_1, T)}{\partial t_1 \partial T} \frac{\partial^2 \Phi_1(t_1, T)}{\partial T \partial t_1} \\ = c_h(a + bT) e^{\theta(T - t_1)} \left[\frac{A}{k_1} e^{-k_1 t_1} + B - C - \frac{c_h b}{\theta} e^{\theta(T - t_1)} \right]$$

Clearly, $|H_{22}|$ is positive when (8) is preserved.

Therefore, $\Phi_1(t_1, T)$ is a positive definite and hence it is a convex function of t_1 and T. Also $\Phi_2(t_1, T)$ is non-negative and differentiable concerning t_1 and T. Furthermore, $\Phi_2(t_1, T)$ is positive and affine function. Consequently, the average cost function $\Phi(t_1, T)$ is strictly a pseudo-convex function in t_1 and T, and it has a unique minimum value.

The first-order partial derivatives of $\Phi(t_1, T)$ concerning t_1 and T equal to zero yield the necessary

conditions for minimizing the total cost $\Phi(t_1, T)$.

$$\frac{\partial \Phi(t_1, T)}{\partial t_1} = \frac{1}{T} \left[\frac{(c_h - nc_p)(k_1k_3 + k_2)}{k_1^2} (1 - e^{-k_1t_1}) + c_p(m + al) + \left(\frac{c_hb}{\theta} + c_plb\right) t_1 - (c_h - nc_p)\frac{k_2}{k_1} t_1 + \frac{c_h}{\theta^2} \{b - \theta(a + bT)\} e^{\theta(T - t_1)} \right] = 0$$

which gives

$$\frac{(c_h - nc_p)(k_1k_3 + k_2)}{k_1^2} (1 - e^{-k_1t_1}) + c_p(m + al) + \left(\frac{c_hb}{\theta} + c_plb\right) t_1 - (c_h - nc_p)\frac{k_2}{k_1} t_1 + \frac{c_h}{\theta^2} \{b - \theta(a + bT)\} e^{\theta(T - t_1)} = 0$$
(9)

and

$$\frac{\partial \Phi(t_1,T)}{\partial T} = -\frac{\Phi_1(t_1,T)}{T^2} + \frac{\partial \Phi_1(t_1,T)}{\partial T} = 0$$

which gives

$$C_{0} + \frac{(c_{h} - nc_{p})(k_{1}k_{3} + k_{2})}{k_{1}^{3}}(e^{-k_{1}t_{1}} + t_{1}k_{1} - 1) + c_{p}(m + al)t_{1} + \left(\frac{c_{h}b}{\theta} + c_{p}lb\right)\frac{t_{1}^{2}}{2} - (c_{h} - nc_{p})$$
$$\times \frac{k_{2}t_{1}^{2}}{2k_{1}} - \frac{c_{h}b}{2\theta}T^{2} + \frac{c_{h}}{\theta^{3}}\{b - \theta(a + bT)\}\{1 - e^{\theta(T - t_{1})}\} + \frac{c_{h}T}{\theta^{2}}\{b - \theta(a + bT)e^{\theta(T - t_{1})}\} = 0 \quad (10)$$

6. Proposed EPQ model in a neutrosophic environment

In this subsection, we reconstruct the manufacturing-supply scenario in the neutrosophic arena, letting three parameters \tilde{a} , \tilde{b} and $\tilde{\theta}$ as neutrosophic numbers. Then, the neutrosophic counterpart of the equations (1) and (2) is given as follows:

For the productive phase $(0 \le t \le t_1)$:

$$\begin{cases} \frac{d\tilde{q}(t)}{dt} = m - (1-l)\tilde{a} - (1-l)\tilde{b}t - (\tilde{\theta}+n)\tilde{q}(t) \\ with \ \tilde{q}(0) = 0, \ \tilde{q}(t_1) = \tilde{Q} \end{cases}$$
(11)

For the non-productive phase $(t_1 \le t \le T)$:

$$\begin{cases} \frac{d\tilde{q}(t)}{dt} = -\{\tilde{a} + \tilde{b}t\} - \tilde{\theta}\tilde{q}(t) \\ with \ \tilde{q}(t_1) = \tilde{Q}, \ \tilde{q}(T) = 0 \end{cases}$$
(12)

Now the concept of generalized neutrosophic differentiation is applied to explain the neutrosophic differential equations (11) and (12). Suppose (α, β, γ) -cut of the neutrosophic valued function $\tilde{q}(t)$ is given as

$$\left[\tilde{q}(t)\right]_{(\alpha,\beta,\gamma)} = \left\langle [q_1(t;\alpha), q_2(t;\alpha)], [q_1'(t;\beta), q_2'(t;\beta)], [q_1''(t;\gamma), q_2''(t;\gamma)] \right\rangle$$

Also, let the parametric representations, i.e., (α, β, γ) -cut of the neutrosophic number \tilde{a} , \tilde{b} and $\tilde{\theta}$ be given as

$$\begin{split} & \left[\hat{\theta}\right]_{(\alpha,\beta,\gamma)} = \langle \left[\theta_1(\alpha), \theta_2(\alpha)\right], \left[\theta_1'(\beta), \theta_2'(\beta)\right], \left[\theta_1''(\gamma), \theta_2''(\gamma)\right] \rangle \\ & \left[\tilde{a}\right]_{(\alpha,\beta,\gamma)} = \langle \left[a_1(\alpha), a_2(\alpha)\right], \left[a_1'(\beta), a_2'(\beta)\right], \left[a_1''(\gamma), a_2''(\gamma)\right] \rangle \\ & \left[\tilde{b}\right]_{(\alpha,\beta,\gamma)} = \langle \left[b_1(\alpha), b_2(\alpha)\right], \left[b_1'(\beta), b_2'(\beta)\right], \left[b_1''(\gamma), b_2''(\gamma)\right] \rangle \end{split}$$

The following two cases are considered according to the two types of generalized neutrosophic derivatives of the neutrosophic valued function $\tilde{q}(t)$.

Case 1: When $\tilde{q}(t)$ is type-1 neutrosophic differentiable

Then, the differential equation (11) represents the productive phase $(0 \le t \le t_1)$ and is turned into a parametric form as

$$\begin{split} \langle [\dot{q}_{1}(t;\alpha),\dot{q}_{2}(t;\alpha)], [\dot{q}_{1}'(t;\beta),\dot{q}_{2}'(t;\beta)], [\dot{q}_{1}''(t;\gamma),\dot{q}_{2}''(t;\gamma)] \rangle &= m - (1-l) \langle [a_{1}(\alpha),a_{2}(\alpha)], \\ [a_{1}'(\beta),a_{2}'(\beta)], [a_{1}''(\gamma),a_{2}''(\gamma)] \rangle - (1-l) \langle [b_{1}(\alpha),b_{2}(\alpha)], [b_{1}'(\beta),b_{2}'(\beta)], [b_{1}''(\gamma),b_{2}''(\gamma)] \rangle t \\ &- \big\{ \langle [\theta_{1}(\alpha),\theta_{2}(\alpha)], [\theta_{1}'(\beta),\theta_{2}'(\beta)], [\theta_{1}''(\gamma),\theta_{2}''(\gamma)] \rangle + n \big\} \langle [q_{1}(t;\alpha),q_{2}(t;\alpha)], \\ & [q_{1}'(t;\beta),q_{2}'(t;\beta)], [q_{1}''(t;\gamma),q_{2}''(t;\gamma)] \rangle \end{split}$$

The above expression gives a system of crisp differential equations as follows:

$$\dot{q}_1(t;\alpha) = m - (1-l)a_2(\alpha) - (1-l)b_2(\alpha)t - \{\theta_2(\alpha) + n\}q_2(t;\alpha)$$
(13)

$$\dot{q}_2(t;\alpha) = m - (1-l)a_1(\alpha) - (1-l)b_1(\alpha)t - \{\theta_1(\alpha) + n\}q_1(t;\alpha)$$
(14)

$$\dot{q}_1'(t;\beta) = m - (1 - l)a_2'(\beta) - (1 - l)b_2'(\beta)t - \{\theta_2'(\beta) + n\}q_2'(t;\beta)$$
(15)

$$\dot{q}_{2}'(t;\beta) = m - (1-l)a_{1}'(\beta) - (1-l)b_{1}'(\beta)t - \{\theta_{1}'(\beta) + n\}q_{1}'(t;\beta)$$
(16)

$$\dot{q}_{1}''(t;\gamma) = m - (1-l)a_{2}''(\gamma) - (1-l)b_{2}''(\gamma)t - \{\theta_{2}''(\gamma) + n\}q_{2}''(t;\gamma)$$
(17)

$$\dot{q}_{2}''(t;\gamma) = m - (1-l)a_{1}''(\gamma) - (1-l)b_{1}''(\gamma)t - \{\theta_{1}''(\gamma) + n\}q_{1}''(t;\gamma)$$
(18)

with $q_1(0; \alpha) = q_2(0; \alpha) = q'_1(0; \beta) = q'_2(0; \beta) = q''_1(0; \gamma) = q''_2(0; \gamma) = 0$. Equations (13) and (14) can be simplified as

$$\begin{cases} \dot{q}_1(t;\alpha) = -u_2 q_2(t;\alpha) - v_2 t + w_2 \\ \dot{q}_2(t;\alpha) = -u_1 q_1(t;\alpha) - v_1 t + w_1 \end{cases}$$
(19)

where $u_2 = \theta_1(\alpha) + n$, $u_1 = \theta_2(\alpha) + n$, $v_1 = (1 - l)b_1(\alpha)$, $v_2 = (1 - l)b_2(\alpha)$, $w_1 = m - (1 - l)a_1(\alpha)$ and $w_2 = m - (1 - l)a_2(\alpha)$.

The solution of the system (19) is given by

$$\begin{cases} q_1(t;\alpha) = c_1 e^{\sqrt{u_1 u_2}t} + c_2 e^{-\sqrt{u_1 u_2}t} + \frac{u_2(w_1 - v_1t) + v_2}{u_1 u_2} \\ q_2(t;\alpha) = -\sqrt{\frac{u_1}{u_2}} c_1 e^{\sqrt{u_1 u_2}t} + \sqrt{\frac{u_1}{u_2}} c_2 e^{-\sqrt{u_1 u_2}t} + \frac{u_1(w_2 - v_2t) + v_1}{u_1 u_2} \end{cases}$$
(20)

Using the initial information $q_1(0; \alpha) = 0$ and $q_2(0; \alpha) = 0$, the values of the constants c_1 and c_2 are obtained as

$$\begin{cases} c_1 = -\frac{\sqrt{u_1}(u_2w_1 + v_2) - \sqrt{u_2}(u_1w_2 + v_1)}{2\sqrt{u_1}u_1u_2} \\ c_2 = -\frac{\sqrt{u_1}(u_2w_1 + v_2) + \sqrt{u_2}(u_1w_2 + v_1)}{2\sqrt{u_1}u_1u_2} \end{cases}$$
(21)

Similarly, the solution of the remaining system of equations (15)-(16) and (17)-(18) are obtained as

$$\begin{cases} q_1'(t;\beta) = c_3 e^{\sqrt{u_3 u_4 t}} + c_4 e^{-\sqrt{u_3 u_4 t}} + \frac{u_4(w_3 - v_3 t) + v_4}{u_3 u_4} \\ q_2'(t;\beta) = -\sqrt{\frac{u_3}{u_4}} c_3 e^{\sqrt{u_3 u_4 t}} + \sqrt{\frac{u_3}{u_4}} c_4 e^{-\sqrt{u_3 u_4 t}} + \frac{u_3(w_4 - v_4 t) + v_3}{u_3 u_4} \\ q_1''(t;\gamma) = c_5 e^{\sqrt{u_5 u_6 t}} + c_6 e^{-\sqrt{u_5 u_6 t}} + \frac{u_6(w_5 - v_5 t) + v_6}{u_5 u_6} \\ q_2''(t;\gamma) = -\sqrt{\frac{u_5}{u_6}} c_5 e^{\sqrt{u_5 u_6 t}} + \sqrt{\frac{u_5}{u_6}} c_6 e^{-\sqrt{u_5 u_6 t}} + \frac{u_5(w_6 - v_6 t) + v_5}{u_5 u_6} \end{cases}$$
(22)

where u_i , v_i and w_i (i = 3, 4, 5, 6) are taken as follows

$$\begin{cases} u_3 = \theta'_1(\beta) + n, \ v_3 = b'_1(\beta)(1-l), \ w_3 = m + a'_1(\beta)(l-1) \\ u_4 = \theta'_2(\beta) + n, \ v_4 = b'_2(\beta)(1-l), \ w_4 = m + a'_2(\beta)(l-1) \\ u_5 = \theta''_1(\gamma) + n, \ v_5 = b''_1(\gamma)(1-l), \ w_5 = m + a''_1(\gamma)(l-1) \\ u_6 = \theta''_2(\gamma) + n, \ v_6 = b''_2(\gamma)(1-l), \ w_6 = m + a''_2(\gamma)(l-1) \end{cases}$$

and the constants are c_3 , c_4 , c_5 and c_4 are given as

$$\begin{cases} c_3 = -\frac{\sqrt{u_3}(u_4w_3 + v_4) - \sqrt{u_4}(u_3w_4 + v_3)}{2\sqrt{u_3}u_3u_4} \\ c_4 = -\frac{\sqrt{u_3}(u_4w_3 + v_4) + \sqrt{u_4}(u_3w_4 + v_3)}{2\sqrt{u_3}u_3u_4} \\ c_5 = -\frac{\sqrt{u_5}(u_6w_5 + v_6) - \sqrt{u_6}(u_5w_6 + v_5)}{2\sqrt{u_5}u_5u_6} \\ c_6 = -\frac{\sqrt{u_5}(u_6w_5 + v_6) + \sqrt{u_6}(u_5w_6 + v_5)}{2\sqrt{u_5}u_5u_6} \end{cases}$$

Again, the neutrosophic differential equation (12) represents the non-productive phase $(t_1 \le t \le T)$ is turned in the parametric form,

$$\langle [\dot{q}_{1}(t;\alpha), \dot{q}_{2}(t;\alpha)], [\dot{q}_{1}'(t;\beta), \dot{q}_{2}'(t;\beta)], [\dot{q}_{1}''(t;\gamma), \dot{q}_{2}''(t;\gamma)] \rangle = -\langle [a_{1}(\alpha), a_{2}(\alpha)], [a_{1}'(\beta), a_{2}'(\beta)], \\ [a_{1}''(\gamma), a_{2}''(\gamma)] \rangle - \langle [b_{1}(\alpha), b_{2}(\alpha)], [b_{1}'(\beta), b_{2}'(\beta)], [b_{1}''(\gamma), b_{2}''(\gamma)] \rangle t - \langle [\theta_{1}(\alpha), \theta_{2}(\alpha)], \\ [\theta_{1}'(\beta), \theta_{2}'(\beta)], [\theta_{1}''(\gamma), \theta_{2}''(\gamma)] \rangle \langle [q_{1}(t;\alpha), q_{2}(t;\alpha)], [q_{1}'(t;\beta), q_{2}'(t;\beta)], [q_{1}''(t;\gamma), q_{2}''(t;\gamma)] \rangle$$

which gives a system of differential equations as follows:

$$\dot{q}_1(t;\alpha) = -a_2(\alpha) - b_2(\alpha)t - \theta_2(\alpha)q_2(t;\alpha)$$
(23)

$$\dot{q}_2(t;\alpha) = -a_1(\alpha) - b_1(\alpha)t - \theta_1(\alpha)q_1(t;\alpha)$$
(24)

$$\dot{q}_1'(t;\beta) = -a_2'(\beta) - b_2'(\beta)t - \theta_2'(\beta)q_2'(t;\beta)$$
(25)

$$\dot{q}_{2}'(t;\beta) = -a_{1}'(\beta) - b_{1}'(\beta)t - \theta_{1}'(\beta)q_{1}'(t;\beta)$$
(26)

$$\dot{q}_1''(t;\gamma) = -a_2''(\gamma) - b_2''(\gamma)t - \theta_2''(\gamma)q_2''(t;\gamma)$$
(27)

$$\dot{q}_{2}''(t;\gamma) = -a_{1}''(\gamma) - b_{1}''(\gamma)t - \theta_{1}''(\gamma)q_{1}''(t;\gamma)$$
(28)

with $q_1(T; \alpha) = q_2(T; \alpha) = q'_1(T; \beta) = q'_2(T; \beta) = q''_1(T; \gamma) = q''_2(T; \gamma) = 0$. By solving the equations (23) and (24) one can get

$$\begin{cases} q_1(t;\alpha) = c_7 e^{\sqrt{\theta_1(\alpha)\theta_2(\alpha)t}} + c_8 e^{-\sqrt{\theta_1(\alpha)\theta_2(\alpha)t}} + \frac{b_2(\alpha) - \theta_2(\alpha)(b_1(\alpha)t + a_1(\alpha))}{\theta_1(\alpha)\theta_2(\alpha)} \\ q_2(t;\alpha) = c_7 \sqrt{\frac{\theta_1(\alpha)}{\theta_2(\alpha)}} e^{\sqrt{\theta_1(\alpha)\theta_2(\alpha)t}} + c_8 \sqrt{\frac{\theta_1(\alpha)}{\theta_2(\alpha)}} e^{-\sqrt{\theta_1(\alpha)\theta_2(\alpha)t}} + \frac{b_1(\alpha) - \theta_1(\alpha)(b_2(\alpha)t + a_2(\alpha))}{\theta_1(\alpha)\theta_2(\alpha)} \end{cases}$$
(29)

Using the initial information $q_1(T; \alpha) = 0$, $q_2(T; \alpha) = 0$ the values of the constants c_7 and c_8 are obtained as

$$\begin{cases} c_{7} = -\frac{e^{-\sqrt{\theta_{1}(\alpha)\theta_{2}(\alpha)}T}}{2\sqrt{\theta_{1}(\alpha)\theta_{1}(\alpha)\theta_{2}(\alpha)}} \left[\sqrt{\theta_{1}(\alpha)} \left\{ b_{2}(\alpha) - \theta_{2}(\alpha)(b_{1}(\alpha)T + a_{1}(\alpha)) \right\} - \sqrt{\theta_{2}(\alpha)} \left\{ b_{1}(\alpha) - \theta_{1}(\alpha)(b_{2}(\alpha)T + a_{2}(\alpha)) \right\} \right] \\ c_{8} = -\frac{e^{\sqrt{\theta_{1}(\alpha)\theta_{2}(\alpha)}T}}{2\sqrt{\theta_{1}(\alpha)\theta_{1}(\alpha)\theta_{2}(\alpha)}} \left[\sqrt{\theta_{1}(\alpha)} \left\{ b_{2}(\alpha) - \theta_{2}(\alpha)(b_{1}(\alpha)T + a_{1}(\alpha)) \right\} + \sqrt{\theta_{2}(\alpha)} \left\{ b_{1}(\alpha) - \theta_{1}(\alpha)(b_{2}(\alpha)T + a_{2}(\alpha)) \right\} \right]$$
(30)

Similarly, by solving the remaining equations (25)-(28) one can get,

$$\begin{cases} q_{1}'(t;\beta) = c_{9}e^{\sqrt{\theta_{1}'(\beta)\theta_{2}'(\beta)t}} + c_{10}e^{-\sqrt{\theta_{1}'(\beta)\theta_{2}'(\beta)t}} + \frac{b_{2}'(\beta)-\theta_{2}'(\beta)(b_{1}'(\beta)t+a_{1}'(\beta))}{\theta_{1}'(\beta)\theta_{2}'(\beta)} \\ q_{2}'(t;\beta) = c_{9}\sqrt{\frac{\theta_{1}'(\beta)}{\theta_{2}'(\beta)}}e^{\sqrt{\theta_{1}'(\beta)\theta_{2}'(\beta)t}} + c_{10}\sqrt{\frac{\theta_{1}'(\beta)}{\theta_{2}'(\beta)}}e^{-\sqrt{\theta_{1}'(\beta)\theta_{2}'(\beta)t}} + \frac{b_{1}'(\beta)-\theta_{1}'(\beta)(b_{2}'(\beta)t+a_{2}'(\beta))}{\theta_{1}'(\beta)\theta_{2}'(\beta)} \\ q_{1}''(t;\gamma) = c_{11}e^{\sqrt{\theta_{1}''(\gamma)\theta_{2}''(\gamma)t}} + c_{12}e^{-\sqrt{\theta_{1}''(\gamma)\theta_{2}''(\gamma)t}} + \frac{b_{2}'(\gamma)-\theta_{2}''(\gamma)(b_{1}''(\gamma)t+a_{1}''(\gamma))}{\theta_{1}'(\gamma)\theta_{2}''(\gamma)t}} \\ q_{2}''(t;\gamma) = c_{11}\sqrt{\frac{\theta_{1}''(\gamma)}{\theta_{2}''(\gamma)}}e^{\sqrt{\theta_{1}''(\gamma)\theta_{2}''(\gamma)t}} + c_{12}\sqrt{\frac{\theta_{1}''(\gamma)}{\theta_{2}''(\gamma)}}e^{-\sqrt{\theta_{1}''(\gamma)\theta_{2}''(\gamma)t}} + \frac{b_{1}''(\gamma)-\theta_{1}''(\gamma)(b_{2}''(\gamma)t+a_{2}''(\gamma))}{\theta_{1}''(\gamma)\theta_{2}''(\gamma)}} \end{cases}$$
(31)

Where the values of the constants c_9, c_{10}, c_{11} and c_{12} are obtained as

$$c_{9} = -\frac{e^{-\sqrt{\theta_{1}'(\beta)\theta_{2}'(\beta)T}}}{2\sqrt{\theta_{1}'(\beta)\theta_{1}'(\beta)\theta_{2}'(\beta)}} \left[\sqrt{\theta_{1}'(\beta)} \{ b_{2}'(\beta) - \theta_{2}'(\beta)(b_{1}'(\beta)T + a_{1}'(\beta)) \} - \sqrt{\theta_{2}'(\beta)} \{ b_{1}'(\beta) - \theta_{1}'(\beta)(b_{2}'(\beta)T + a_{2}'(\beta)) \} \right]$$

$$c_{10} = -\frac{e^{\sqrt{\theta_{1}'(\beta)\theta_{2}'(\beta)T}}}{2\sqrt{\theta_{1}'(\beta)\theta_{1}'(\beta)\theta_{2}'(\beta)}} \left[\sqrt{\theta_{1}'(\beta)} \{ b_{2}'(\beta) - \theta_{2}'(\beta)(b_{1}'(\beta)T + a_{1}'(\beta)) \} + \sqrt{\theta_{2}'(\beta)} \{ b_{1}'(\beta) - \theta_{1}'(\beta)(b_{2}'(\beta)T + a_{2}'(\beta)) \} \right]$$

$$c_{11} = -\frac{e^{-\sqrt{\theta_{1}''(\gamma)\theta_{2}'(\gamma)T}}}{2\sqrt{\theta_{1}''(\gamma)\theta_{1}''(\gamma)\theta_{2}''(\gamma)}} \left[\sqrt{\theta_{1}''(\gamma)} \{ b_{2}''(\gamma) - \theta_{2}''(\gamma)(b_{1}''(\gamma)T + a_{1}''(\gamma)) \} - \sqrt{\theta_{2}''(\gamma)} \{ b_{1}''(\gamma) - \theta_{1}''(\gamma)(b_{2}''(\gamma)T + a_{2}''(\gamma)) \} \right]$$

$$c_{12} = -\frac{e\sqrt{\theta_{1}''(\gamma)\theta_{1}''(\gamma)\theta_{1}''(\gamma)\theta_{2}''(\gamma)T}}{2\sqrt{\theta_{1}''(\gamma)\theta_{1}''(\gamma)\theta_{2}''(\gamma)T}} \left[\sqrt{\theta_{1}''(\gamma)} \{ b_{2}''(\gamma) - \theta_{2}''(\gamma)(b_{1}''(\gamma)T + a_{1}''(\gamma)) \} + \sqrt{\theta_{2}''(\gamma)} \{ b_{1}''(\gamma) - \theta_{1}''(\gamma)(b_{2}''(\gamma)T + a_{2}''(\gamma)) \} \right]$$

Some relevant costs: Therefore, the holding cost, $\tilde{HC} = \langle [HC_1(\alpha), HC_2(\alpha)], [HC'_1(\beta), HC'_2(\beta)], [HC''_1(\gamma), HC''_2(\gamma)] \rangle$ given by $HC(\alpha) = \int_{-\infty}^{\infty} \int_{-\infty}^{0} f^{t_1} dt = \int_{-\infty}^{0} \int_{-\infty}^{0} \int_{-\infty}^{0} \int_{-\infty}^{0} f^{t_1} dt = \int_{-\infty}^{0} \int_{-$

$$\begin{split} HC_{1}(\alpha) &= c_{h} \bigg[\int_{0}^{t_{1}} q_{1}(t;\alpha) dt + \int_{t_{1}}^{T} q_{2}(t;\alpha) dt \bigg] \\ &= c_{h} \bigg[\int_{0}^{t_{1}} \bigg\{ c_{1} e^{\sqrt{u_{1}u_{2}t}} + c_{2} e^{-\sqrt{u_{1}u_{2}t}} + \frac{u_{2}(w_{1} - v_{1}t) + v_{2}}{u_{1}u_{2}} \bigg\} dt + \int_{t_{1}}^{T} \bigg\{ c_{7} e^{\sqrt{\theta_{1}(\alpha)\theta_{2}(\alpha)t}} \\ &+ c_{8} e^{-\sqrt{\theta_{1}(\alpha)\theta_{2}(\alpha)t}} + \frac{b_{2}(\alpha) - \theta_{2}(\alpha)(b_{1}(\alpha)t + a_{1}(\alpha))}{\theta_{1}(\alpha)\theta_{2}(\alpha)} \bigg\} dt \bigg] \\ &= c_{h} \bigg[\bigg\{ \frac{c_{1}}{\sqrt{u_{1}u_{2}}} \Big(e^{\sqrt{u_{1}u_{2}t_{1}}} - 1 \Big) - \frac{c_{2}}{\sqrt{u_{1}u_{2}}} \Big(e^{-\sqrt{u_{1}u_{2}t_{1}}} - 1 \Big) + \frac{2(u_{2}w_{1} + v_{2})t_{1} - u_{2}v_{1}t_{1}^{2}}{2u_{1}u_{2}} \bigg\} \\ &+ \bigg\{ \frac{c_{7}}{\sqrt{\theta_{1}(\alpha)\theta_{2}(\alpha)}} \Big(e^{\sqrt{\theta_{1}(\alpha)\theta_{2}(\alpha)T}} - e^{\sqrt{\theta_{1}(\alpha)\theta_{2}(\alpha)t_{1}}} \Big) + \frac{c_{8}}{\sqrt{\theta_{1}(\alpha)\theta_{2}(\alpha)}} \bigg(e^{-\sqrt{\theta_{1}(\alpha)\theta_{2}(\alpha)T}} \\ &- e^{-\sqrt{\theta_{1}(\alpha)\theta_{2}(\alpha)t_{1}}} \bigg) - \frac{\theta_{2}(\alpha)b_{1}(\alpha)(T^{2} - t_{1}^{2}) + 2(\theta_{2}(\alpha)a_{1}(\alpha) - b_{2}(\alpha))(T - t_{1})}{2\theta_{1}(\alpha)\theta_{2}(\alpha)} \bigg\} \bigg] \end{split}$$

Similarly, solving the other components of the holding cost as

$$HC_{2}(\alpha) = c_{h} \left[\left\{ \frac{c_{1}}{u_{2}} \left(1 - e^{\sqrt{u_{1}u_{2}}t_{1}} \right) - \frac{c_{2}}{u_{2}} \left(1 - e^{-\sqrt{u_{1}u_{2}}t_{1}} \right) + \frac{2(u_{1}w_{2} + v_{1})t_{1} - u_{1}v_{2}t_{1}^{2}}{2u_{1}u_{2}} \right) \right] \\ + \left\{ \frac{c_{7}}{\theta_{2}(\alpha)} \left(e^{\sqrt{\theta_{1}(\alpha)\theta_{2}(\alpha)}t_{1}} - e^{\sqrt{\theta_{1}(\alpha)\theta_{2}(\alpha)}T} \right) + \frac{c_{8}}{\theta_{2}(\alpha)} \left(e^{-\sqrt{\theta_{1}(\alpha)\theta_{2}(\alpha)}t_{1}} - e^{-\sqrt{\theta_{1}(\alpha)\theta_{2}(\alpha)}T} \right) - \frac{\theta_{1}(\alpha)b_{2}(\alpha)(T^{2} - t_{1}^{2}) + 2(\theta_{1}(\alpha)a_{2}(\alpha) - b_{1}(\alpha))(T - t_{1})}{2\theta_{1}(\alpha)\theta_{2}(\alpha)} \right\} \right]$$

$$HC_{1}'(\beta) = c_{h} \left[\left\{ \frac{c_{3}}{\sqrt{u_{3}u_{4}}} \left(e^{\sqrt{u_{3}u_{4}}t_{1}} - 1 \right) - \frac{c_{4}}{\sqrt{u_{3}u_{4}}} \left(e^{-\sqrt{u_{3}u_{4}}t_{1}} - 1 \right) + \frac{2(u_{4}w_{3} + v_{4})t_{1} - u_{4}v_{3}t_{1}^{2}}{2u_{3}u_{4}} \right\} \right] \\ + \left\{ \frac{c_{9}}{\sqrt{\theta_{1}'(\beta)\theta_{2}'(\beta)}} \left(e^{\sqrt{\theta_{1}'(\beta)\theta_{2}'(\beta)}T} - e^{\sqrt{\theta_{1}'(\beta)\theta_{2}'(\beta)}t_{1}} \right) + \frac{c_{10}}{\sqrt{\theta_{1}'(\beta)\theta_{2}'(\beta)}} \left(e^{-\sqrt{\theta_{1}'(\beta)\theta_{2}'(\beta)}T} - e^{-\sqrt{\theta_{1}'(\beta)\theta_{2}'(\beta)}t_{1}} \right) - \frac{\theta_{2}'(\beta)b_{1}'(\beta)(T^{2} - t_{1}^{2}) + 2(\theta_{2}'(\beta)a_{1}'(\beta) - b_{2}'(\beta))(T - t_{1})}{2\theta_{1}'(\beta)\theta_{2}'(\beta)} \right\} \right]$$

$$HC'_{2}(\beta) = c_{h} \left[\left\{ \frac{c_{3}}{u_{4}} \left(1 - e^{\sqrt{u_{3}u_{4}}t_{1}} \right) - \frac{c_{4}}{u_{4}} \left(1 - e^{-\sqrt{u_{3}u_{4}}t_{1}} \right) + \frac{2(u_{3}w_{4} + v_{3})t_{1} - u_{3}v_{4}t_{1}^{2}}{2u_{3}u_{4}} \right\} \right. \\ \left. + \left\{ \frac{c_{9}}{\theta'_{2}(\beta)} \left(e^{\sqrt{\theta'_{1}(\beta)\theta'_{2}(\beta)}t_{1}} - e^{\sqrt{\theta'_{1}(\beta)\theta'_{2}(\beta)}T} \right) + \frac{c_{10}}{\theta'_{2}(\beta)} \left(e^{-\sqrt{\theta'_{1}(\beta)\theta'_{2}(\beta)}t_{1}} - e^{-\sqrt{\theta'_{1}(\beta)\theta'_{2}(\beta)}T} \right) - \frac{\theta'_{1}(\beta)b'_{2}(\beta)(T^{2} - t_{1}^{2}) + 2(\theta'_{1}(\beta)a'_{2}(\beta) - b'_{1}(\beta))(T - t_{1})}{2\theta'_{1}(\beta)\theta'_{2}(\beta)} \right\} \right]$$

$$HC_{1}''(\gamma) = c_{h} \left[\left\{ \frac{c_{5}}{\sqrt{u_{5}u_{6}}} \left(e^{\sqrt{u_{5}u_{6}}t_{1}} - 1 \right) - \frac{c_{6}}{\sqrt{u_{5}u_{6}}} \left(e^{-\sqrt{u_{5}u_{6}}t_{1}} - 1 \right) + \frac{2(u_{6}w_{5} + v_{6})t_{1} - u_{6}v_{5}t_{1}^{2}}{2u_{5}u_{6}} \right) \right] \right] \\ + \left\{ \frac{c_{11}}{\sqrt{\theta_{1}''(\gamma)\theta_{2}''(\gamma)}} \left(e^{\sqrt{\theta_{1}''(\gamma)\theta_{2}''(\gamma)T}} - e^{\sqrt{\theta_{1}''(\gamma)\theta_{2}''(\gamma)t_{1}}} \right) + \frac{c_{12}}{\sqrt{\theta_{1}''(\gamma)\theta_{2}''(\gamma)}} \left(e^{-\sqrt{\theta_{1}''(\gamma)\theta_{2}''(\gamma)T}} - e^{\sqrt{\theta_{1}''(\gamma)\theta_{2}''(\gamma)t_{1}}} \right) - \frac{\theta_{2}''(\gamma)b_{1}''(\gamma)(T^{2} - t_{1}^{2}) + 2(\theta_{2}''(\gamma)a_{1}''(\gamma) - b_{2}''(\gamma))(T - t_{1})}}{2\theta_{1}''(\gamma)\theta_{2}''(\gamma)}} \right\} \right] \\ HC_{2}''(\gamma) = c_{h} \left[\left\{ \frac{c_{5}}{u_{6}} \left(1 - e^{\sqrt{u_{5}u_{6}}t_{1}} \right) - \frac{c_{6}}{u_{6}} \left(1 - e^{-\sqrt{u_{5}u_{6}}t_{1}} \right) + \frac{2(u_{5}w_{6} + v_{5})t_{1} - u_{5}v_{6}t_{1}^{2}}{2u_{5}u_{6}}} \right\} \\ + \int c_{11} \left(e^{\sqrt{\theta_{1}''(\gamma)\theta_{2}''(\gamma)}t_{1}} - e^{\sqrt{\theta_{1}''(\gamma)\theta_{2}''(\gamma)T}} \right) + \frac{c_{12}}{2(e^{-\sqrt{\theta_{1}''(\gamma)\theta_{2}''(\gamma)}t_{1}})} \right] dt_{1} dt_{1} dt_{2} dt$$

$$+ \left\{ \frac{\partial \Pi}{\partial_{2}''(\gamma)} \left(e^{\sqrt{\theta_{1}'(\gamma)\theta_{2}'(\gamma)t_{1}}} - e^{\sqrt{\theta_{1}'(\gamma)\theta_{2}'(\gamma)T}} \right) + \frac{\partial \Pi}{\partial_{2}''(\gamma)} \left(e^{-\sqrt{\theta_{1}'(\gamma)\theta_{2}'(\gamma)t_{1}}} - e^{-\sqrt{\theta_{1}''(\gamma)\theta_{2}''(\gamma)T}} \right) - \frac{\theta_{1}''(\gamma)b_{2}''(\gamma)(T^{2} - t_{1}^{2}) + 2(\theta_{1}''(\gamma)a_{2}''(\gamma) - b_{1}''(\gamma))(T - t_{1})}{2\theta_{1}''(\gamma)\theta_{2}''(\gamma)} \right\} \right]$$

Therefore, the production cost $\tilde{PC} = \langle [PC_1(\alpha), PC_2(\alpha)], [PC'_1(\beta), PC'_2(\beta)], [PC''_1(\gamma), PC''_2(\gamma)] \rangle$ during the entire circle is given by

$$PC_{1}(\alpha) = c_{p} \left[\int_{0}^{t_{1}} \left\{ (m + la_{1}(\alpha)) - nq_{1}(t;\alpha) + lb_{1}(\alpha)t \right\} dt \right]$$

$$= c_{p} \left[(m + la_{1}(\alpha))t_{1} + lb_{1}(\alpha)\frac{t_{1}^{2}}{2} - n \int_{0}^{t_{1}} q_{1}(t;\alpha) dt \right]$$

$$= c_{p} \left[(m + la_{1}(\alpha))t_{1} + lb_{1}(\alpha)\frac{t_{1}^{2}}{2} - n \left\{ \frac{c_{1}}{\sqrt{u_{1}u_{2}}} \left(e^{\sqrt{u_{1}u_{2}}t_{1}} - 1\right) - \frac{c_{2}}{\sqrt{u_{1}u_{2}}} \left(e^{-\sqrt{u_{1}u_{2}}t_{1}} - 1\right) + \frac{2(u_{2}w_{1} + v_{2})t_{1} - u_{2}v_{1}t_{1}^{2}}{2u_{1}u_{2}} \right\} \right]$$

$$PC_{2}(\alpha) = c_{p} \left[\int_{0}^{t_{1}} \left\{ (m + la_{2}(\alpha)) - nq_{2}(t;\alpha) + lb_{2}(\alpha)t \right\} dt \right]$$

$$= c_{p} \left[(m + la_{2}(\alpha))t_{1} + lb_{2}(\alpha)\frac{t_{1}^{2}}{2} - n \int_{0}^{t_{1}} q_{2}(t;\alpha)dt \right]$$

$$= c_{p} \left[(m + la_{2}(\alpha))t_{1} + lb_{2}(\alpha)\frac{t_{1}^{2}}{2} - n \left\{ \frac{c_{1}}{u_{2}} \left(1 - e^{\sqrt{u_{1}u_{2}t_{1}}} \right) - \frac{c_{2}}{u_{2}} \left(1 - e^{-\sqrt{u_{1}u_{2}t_{1}}} \right) + \frac{2(u_{1}w_{2} + v_{1})t_{1} - u_{1}v_{2}t_{1}^{2}}{2u_{1}u_{2}} \right\} \right]$$

$$PC_{1}'(\beta) = c_{p} \left[(m + la_{1}'(\beta))t_{1} + lb_{1}'(\beta)\frac{t_{1}^{2}}{2} - n \left\{ \frac{c_{3}}{\sqrt{u_{3}u_{4}}} \left(e^{\sqrt{u_{3}u_{4}}t_{1}} - 1 \right) - \frac{c_{4}}{\sqrt{u_{3}u_{4}}} \left(e^{-\sqrt{u_{3}u_{4}}t_{1}} - 1 \right) + \frac{2(u_{4}w_{3} + v_{4})t_{1} - u_{4}v_{3}t_{1}^{2}}{2u_{3}u_{4}} \right\} \right]$$

$$PC_{1}''(\beta) = c_{p} \left[(m + la_{2}'(\beta))t_{1} + lb_{2}'(\beta)\frac{t_{1}^{2}}{2} - n \left\{ \frac{c_{3}}{u_{4}} \left(1 - e^{\sqrt{u_{3}u_{4}}t_{1}} \right) - \frac{c_{4}}{u_{4}} \left(1 - e^{-\sqrt{u_{3}u_{4}}t_{1}} \right) + \frac{2(u_{3}w_{4} + v_{3})t_{1} - u_{3}v_{4}t_{1}^{2}}{2u_{3}u_{4}} \right\} \right]$$

$$PC_{2}''(\gamma) = c_{p} \left[(m + la_{1}''(\gamma))t_{1} + lb_{1}''(\gamma)\frac{t_{1}^{2}}{2} - n \left\{ \frac{c_{5}}{\sqrt{u_{5}u_{6}}} \left(e^{\sqrt{u_{5}u_{6}}t_{1}} - 1\right) - \frac{c_{6}}{\sqrt{u_{5}u_{6}}} \left(e^{-\sqrt{u_{5}u_{6}}t_{1}} - 1\right) + \frac{2(u_{6}w_{5} + v_{6})t_{1} - u_{6}v_{5}t_{1}^{2}}{2u_{5}u_{6}} \right\} \right]$$

$$PC_{2}''(\gamma) = c_{p} \left[(m + la_{2}''(\gamma))t_{1} + lb_{2}''(\gamma)\frac{t_{1}^{2}}{2} - n \left\{ \frac{c_{5}}{u_{6}} \left(1 - e^{\sqrt{u_{5}u_{6}}t_{1}}\right) - \frac{c_{6}}{u_{6}} \left(1 - e^{-\sqrt{u_{5}u_{6}}t_{1}}\right) + \frac{2(u_{5}w_{6} + v_{5})t_{1} - u_{5}v_{6}t_{1}^{2}}{2u_{5}u_{6}} \right\} \right]$$

Therefore, the total average cost of the system during the entire circle can be obtained in the parametric form as $[\tilde{\Phi}]_{\alpha,\beta,\gamma} = \langle [\Phi_1(\alpha), \Phi_2(\alpha)], [\Phi'_1(\beta), \Phi'_2(\beta)], [\Phi''_1(\gamma), \Phi''_2(\gamma)] \rangle$, where $\Phi_1(\alpha) = \frac{HC_1(\alpha) + PC_1(\alpha) + C_0}{T}$, $\Phi_2(\alpha) = \frac{HC_2(\alpha) + PC_2(\alpha) + C_0}{T}$, $\Phi'_1(\beta) = \frac{HC'_1(\beta) + PC'_1(\beta) + C_0}{T}$, $\Phi'_2(\beta) = \frac{HC'_2(\beta) + PC'_2(\beta) + C_0}{T}$, $\Phi''_1(\gamma) = \frac{HC''_1(\gamma) + PC''_1(\gamma) + C_0}{T}$ and $\Phi''_2(\gamma) = \frac{HC''_2(\gamma) + PC''_2(\gamma) + C_0}{T}$.

Therefore, mathematically, the minimization problem concerning the production inventory model in the case of type-1 neutrosophic differentiability of $\tilde{q}(t)$ is obtained as follows:

$$\begin{cases}
Minimize \ \Phi_{1}(\alpha) \\
Minimize \ \Phi_{2}(\alpha) \\
Minimize \ \Phi_{1}'(\beta) \\
Minimize \ \Phi_{2}'(\beta) \\
Minimize \ \Phi_{2}''(\gamma) \\
Minimize \ \Phi_{2}''(\gamma) \\
Subject \ to \ T \ge t_{1} \ge 0 \\
0 \le \alpha, \beta, \gamma \le 1 \ with \ \alpha + \beta + \gamma \le 3
\end{cases}$$
(32)

Case 2: When $\tilde{q}(t)$ is type-2 neutrosophic differentiable

Then, the differential equation (11) representing the productive phase $(0 \le t \le t_1)$ is turned into a parametric form,

$$\begin{split} \langle [\dot{q}_{2}(t;\alpha),\dot{q}_{1}(t;\alpha)], [\dot{q}_{2}'(t;\beta),\dot{q}_{1}'(t;\beta)], [\dot{q}_{2}''(t;\gamma),\dot{q}_{1}''(t;\gamma)] \rangle &= m - (1-l) \langle [a_{1}(\alpha),a_{2}(\alpha)], \\ [a_{1}'(\beta),a_{2}'(\beta)], [a_{1}''(\gamma),a_{2}''(\gamma)] \rangle - (1-l) \langle [b_{1}(\alpha),b_{2}(\alpha)], [b_{1}'(\beta),b_{2}'(\beta)], [b_{1}''(\gamma),b_{2}''(\gamma)] \rangle t \\ &- \big\{ \langle [\theta_{1}(\alpha),\theta_{2}(\alpha)], [\theta_{1}'(\beta),\theta_{2}'(\beta)], [\theta_{1}''(\gamma),\theta_{2}''(\gamma)] \rangle + n \big\} \langle [q_{1}(t;\alpha),q_{2}(t;\alpha)], \\ & [q_{1}'(t;\beta),q_{2}'(t;\beta)], [q_{1}''(t;\gamma),q_{2}''(t;\gamma)] \rangle \end{split}$$

The above expression gives a system of crisp differential equations as follows:

$$\dot{q}_2(t;\alpha) = m - (1-l)a_2(\alpha) - (1-l)b_2(\alpha)t - \{\theta_2(\alpha) + n\}q_2(t;\alpha)$$
(33)

$$\dot{q}_1(t;\alpha) = m - (1-l)a_1(\alpha) - (1-l)b_1(\alpha)t - \{\theta_1(\alpha) + n\}q_1(t;\alpha)$$
(34)

$$\dot{q}_{2}'(t;\beta) = m - (1-l)a_{2}'(\beta) - (1-l)b_{2}'(\beta)t - \{\theta_{2}'(\beta) + n\}q_{2}'(t;\beta)$$
(35)

$$\dot{q}_1'(t;\beta) = m - (1-l)a_1'(\beta) - (1-l)b_1'(\beta)t - \{\theta_1'(\beta) + n\}q_1'(t;\beta)$$
(36)

$$\dot{q}_{2}''(t;\gamma) = m - (1-l)a_{2}''(\gamma) - (1-l)b_{2}''(\gamma)t - \{\theta_{2}''(\gamma) + n\}q_{2}''(t;\gamma)$$
(37)

$$\dot{q}_{1}''(t;\gamma) = m - (1-l)a_{1}''(\gamma) - (1-l)b_{1}''(\gamma)t - \{\theta_{1}''(\gamma) + n\}q_{1}''(t;\gamma)$$
(38)

with $q_1(0;\alpha) = q_2(0;\alpha) = q'_1(0;\beta) = q'_2(0;\beta) = q''_1(0;\gamma) = q''_2(0;\gamma) = 0.$

Now, the equation (34) can be simplified as

$$\begin{cases} \dot{q}_1(t;\alpha) + u_1 q_1(t;\alpha) = w_1 - v_1 t\\ q_1(0;\alpha) = 0 \end{cases}$$
(39)

The solution of equation (39) is obtained as

$$q_1(t;\alpha) = \frac{u_1 w_1 + v_1}{u_1^2} (1 - e^{-u_1 t}) - \frac{v_1}{u_1} t$$
(40)

Proceeding similarly, the solution of the equations (33) and (35)-(38) are obtained as

$$\begin{cases} q_{2}(t;\alpha) = \frac{u_{2}w_{2}+v_{2}}{u_{2}^{2}}(1-e^{-u_{2}t}) - \frac{v_{2}}{u_{2}}t \\ q_{1}'(t;\beta) = \frac{u_{3}w_{3}+v_{3}}{u_{3}^{2}}(1-e^{-u_{3}t}) - \frac{v_{3}}{u_{3}}t \\ q_{2}'(t;\beta) = \frac{u_{4}w_{4}+v_{4}}{u_{4}^{2}}(1-e^{-u_{4}t}) - \frac{v_{4}}{u_{4}}t \\ q_{1}''(t;\gamma) = \frac{u_{5}w_{5}+v_{5}}{u_{5}^{2}}(1-e^{-u_{5}t}) - \frac{v_{5}}{u_{5}}t \\ q_{2}''(t;\gamma) = \frac{u_{6}w_{6}+v_{6}}{u_{6}^{2}}(1-e^{-u_{6}t}) - \frac{v_{6}}{u_{6}}t \end{cases}$$

$$\tag{41}$$

Again, the neutrosophic differential equation (12) represents the non-productive phase $(t_1 \le t \le T)$ in case of type-2 neutrosophic differentiability is turned in the parametric form,

$$\begin{split} \langle [\dot{q}_{2}(t;\alpha),\dot{q}_{1}(t;\alpha)], [\dot{q}_{2}'(t;\beta),\dot{q}_{1}'(t;\beta)], [\dot{q}_{2}''(t;\gamma),\dot{q}_{1}''(t;\gamma)] \rangle &= -\langle [a_{1}(\alpha),a_{2}(\alpha)], [a_{1}'(\beta),a_{2}'(\beta)], \\ [a_{1}''(\gamma),a_{2}''(\gamma)] \rangle - \langle [b_{1}(\alpha),b_{2}(\alpha)], [b_{1}'(\beta),b_{2}'(\beta)], [b_{1}''(\gamma),b_{2}''(\gamma)] \rangle t - \langle [\theta_{1}(\alpha),\theta_{2}(\alpha)], \\ [\theta_{1}'(\beta),\theta_{2}'(\beta)], [\theta_{1}''(\gamma),\theta_{2}''(\gamma)] \rangle \langle [q_{1}(t;\alpha),q_{2}(t;\alpha)], [q_{1}'(t;\beta),q_{2}'(t;\beta)], [q_{1}''(t;\gamma),q_{2}''(t;\gamma)] \rangle \end{split}$$

which gives a system of differential equations as follows:

$$\dot{q}_2(t;\alpha) = -a_2(\alpha) - b_2(\alpha)t - \theta_2(\alpha)q_2(t;\alpha)$$

$$\tag{42}$$

$$a_1(t;\alpha) = -a_1(\alpha) - b_1(\alpha)t - \theta_1(\alpha)q_1(t;\alpha)$$

$$\tag{43}$$

$$\dot{q}_{2}(t;\alpha) = -a_{2}(\alpha) - b_{2}(\alpha)t - \theta_{2}(\alpha)q_{2}(t;\alpha)$$

$$\dot{q}_{1}(t;\alpha) = -a_{1}(\alpha) - b_{1}(\alpha)t - \theta_{1}(\alpha)q_{1}(t;\alpha)$$

$$\dot{q}'_{2}(t;\beta) = -a'_{2}(\beta) - b'_{2}(\beta)t - \theta'_{2}(\beta)q'_{2}(t;\beta)$$

$$\dot{q}'_{2}(t;\beta) = -a'_{1}(\beta) - b'_{1}(\beta)t - \theta'_{1}(\beta)q'_{1}(t;\beta)$$

$$\dot{q}''_{2}(t;\gamma) = -a''_{2}(\gamma) - b''_{2}(\gamma)t - \theta''_{2}(\gamma)q''_{2}(t;\gamma)$$

$$(42)$$

$$(42)$$

$$(43)$$

$$(43)$$

$$(44)$$

$$(44)$$

$$(45)$$

$$\dot{q}''_{2}(t;\gamma) = -a''_{2}(\gamma) - b''_{2}(\gamma)t - \theta''_{2}(\gamma)q''_{2}(t;\gamma)$$

$$(46)$$

$$\dot{q}_{2}'(t;\beta) = -a_{1}'(\beta) - b_{1}'(\beta)t - \theta_{1}'(\beta)q_{1}'(t;\beta)$$
(45)

$$\dot{q}_{2}''(t;\gamma) = -a_{2}''(\gamma) - b_{2}''(\gamma)t - \theta_{2}''(\gamma)q_{2}''(t;\gamma)$$
(46)

$$\dot{q}_{2}''(t;\gamma) = -a_{1}''(\gamma) - b_{1}''(\gamma)t - \theta_{1}''(\gamma)q_{1}''(t;\gamma)$$
(47)

with $q_1(T; \alpha) = q_2(T; \alpha) = q'_1(T; \beta) = q'_2(T; \beta) = q''_1(T; \gamma) = q''_2(T; \gamma) = 0.$

The solution of the equation (43) is obtained as

$$q_1(t;\alpha) = \frac{b_1(\alpha) - \theta_1(\alpha)(b_1(\alpha)t + a_1(\alpha))}{(\theta_1(\alpha))^2} - \frac{b_1(\alpha) - \theta_1(\alpha)(b_1(\alpha)T + a_1(\alpha))}{(\theta_1(\alpha))^2}e^{\theta_1(\alpha)(T-t_1)}$$
(48)

Proceeding similarly, the solution of the equations (42) and (44)-(47) for the non-productive phase ($t_1 \leq$ $t \leq T$) are obtained as

$$\begin{cases} q_{1}(t;\alpha) = \frac{b_{2}(\alpha)-\theta_{2}(\alpha)(b_{2}(\alpha)t+a_{2}(\alpha))}{(\theta_{2}(\alpha))^{2}} - \frac{b_{2}(\alpha)-\theta_{2}(\alpha)(b_{2}(\alpha)T+a_{2}(\alpha))}{(\theta_{2}(\alpha))^{2}}e^{\theta_{2}(\alpha)(T-t_{1})} \\ q_{1}'(t;\beta) = \frac{b_{1}'(\beta)-\theta_{1}'(\beta)(b_{1}'(\beta)t+a_{1}'(\beta))}{(\theta_{1}'(\beta))^{2}} - \frac{b_{1}'(\beta)-\theta_{1}'(\beta)(b_{1}'(\beta)T+a_{1}'(\beta))}{(\theta_{1}'(\beta))^{2}}e^{\theta_{1}'(\beta)(T-t_{1})} \\ q_{2}'(t;\beta) = \frac{b_{2}'(\beta)-\theta_{2}'(\beta)(b_{2}'(\beta)t+a_{2}'(\beta))}{(\theta_{2}'(\beta))^{2}} - \frac{b_{2}'(\beta)-\theta_{2}'(\beta)(b_{2}'(\beta)T+a_{2}'(\beta))}{(\theta_{2}'(\beta))^{2}}e^{\theta_{2}'(\beta)(T-t_{1})} \\ q_{1}''(t;\gamma) = \frac{b_{1}''(\gamma)-\theta_{1}''(\gamma)(b_{1}''(\gamma)t+a_{1}''(\gamma))}{(\theta_{1}''(\gamma))^{2}} - \frac{b_{1}''(\gamma)-\theta_{1}''(\gamma)(b_{1}''(\gamma)T+a_{1}''(\gamma))}{(\theta_{1}''(\gamma))^{2}}e^{\theta_{1}''(\gamma)(T-t_{1})} \\ q_{2}''(t;\gamma) = \frac{b_{2}'(\gamma)-\theta_{2}''(\gamma)(b_{2}'(\gamma)t+a_{2}''(\gamma))}{(\theta_{2}'(\gamma))^{2}} - \frac{b_{2}'(\gamma)-\theta_{2}''(\gamma)(b_{2}''(\gamma)T+a_{2}''(\gamma))}{(\theta_{1}''(\gamma))^{2}}e^{\theta_{2}''(\gamma)(T-t_{1})} \end{cases}$$
(49)

Several relevant costs associated with the model are formulated as follows: Therefore, the holding cost, $\tilde{HC} = \langle [HC_1(\alpha), HC_2(\alpha)], [HC'_1(\beta), HC'_2(\beta)], [HC''_1(\gamma), HC''_2(\gamma)] \rangle$ given by

$$\begin{split} HC_{1}(\alpha) &= c_{h} \bigg[\int_{0}^{t_{1}} q_{1}(t;\alpha) dt + \int_{t_{1}}^{T} q_{2}(t;\alpha) dt \bigg] \\ &= c_{h} \bigg[\int_{0}^{t_{1}} \bigg\{ \frac{u_{1}w_{1} + v_{1}}{u_{1}^{2}} (1 - e^{-u_{1}t}) - \frac{v_{1}}{u_{1}} t \bigg\} dt + \int_{t_{1}}^{T} \bigg\{ \frac{b_{1}(\alpha) - \theta_{1}(\alpha)(b_{1}(\alpha)t + a_{1}(\alpha))}{(\theta_{1}(\alpha))^{2}} \\ &- \frac{b_{1}(\alpha) - \theta_{1}(\alpha)(b_{1}(\alpha)T + a_{1}(\alpha))}{(\theta_{1}(\alpha))^{2}} e^{\theta_{1}(\alpha)(T - t_{1})} \bigg\} dt \bigg] \\ &= c_{h} \bigg[\frac{u_{1}w_{1} + v_{1}}{u_{1}^{3}} \big(e^{-u_{1}t_{1}} + u_{1}t_{1} - 1 \big) - \frac{v_{1}}{2u_{1}} t_{1}^{2} - \frac{b_{1}(\alpha)}{2\theta_{1}(\alpha)} (T^{2} - t_{1}^{2}) \\ &+ \frac{b_{1}(\alpha) - \theta_{1}(\alpha)(b_{1}(\alpha)T + a_{1}(\alpha))}{(\theta_{1}(\alpha))^{3}} \bigg\{ 1 - e^{\theta_{1}(\alpha)(T - t_{1})} \bigg\} \bigg] \end{split}$$

Similarly, calculating the other components of the holding cost as

$$\begin{aligned} HC_{2}(\alpha) &= c_{h} \bigg[\frac{u_{2}w_{2} + v_{2}}{u_{2}^{3}} \big(e^{-u_{2}t_{1}} + u_{2}t_{1} - 1 \big) - \frac{v_{2}}{2u_{2}} t_{1}^{2} - \frac{b_{2}(\alpha)}{2\theta_{2}(\alpha)} (T^{2} - t_{1}^{2}) \\ &+ \frac{b_{2}(\alpha) - \theta_{2}(\alpha)(b_{2}(\alpha)T + a_{2}(\alpha))}{(\theta_{2}(\alpha))^{3}} \Big\{ 1 - e^{\theta_{2}(\alpha)(T - t_{1})} \Big\} \bigg] \\ HC_{1}'(\beta) &= c_{h} \bigg[\frac{u_{3}w_{3} + v_{3}}{u_{3}^{3}} \big(e^{-u_{3}t_{1}} + u_{3}t_{1} - 1 \big) - \frac{v_{3}}{2u_{3}} t_{1}^{2} - \frac{b_{1}'(\beta)}{2\theta_{1}'(\beta)} (T^{2} - t_{1}^{2}) \\ &+ \frac{b_{1}'(\beta) - \theta_{1}'(\beta)(b_{1}'(\beta)T + a_{1}'(\beta))}{(\theta_{1}'(\beta))^{3}} \Big\{ 1 - e^{\theta_{1}'(\beta)(T - t_{1})} \Big\} \bigg] \end{aligned}$$

$$HC'_{2}(\beta) = c_{h} \left[\frac{u_{4}w_{4} + v_{4}}{u_{4}^{3}} \left(e^{-u_{4}t_{1}} + u_{4}t_{1} - 1 \right) - \frac{v_{4}}{2u_{4}} t_{1}^{2} - \frac{b'_{2}(\beta)}{2\theta'_{2}(\beta)} (T^{2} - t_{1}^{2}) + \frac{b'_{2}(\beta) - \theta'_{2}(\beta)(b'_{2}(\beta)T + a'_{2}(\beta))}{(\theta'_{2}(\beta))^{3}} \left\{ 1 - e^{\theta'_{2}(\beta)(T - t_{1})} \right\} \right]$$

$$HC_{1}''(\gamma) = c_{h} \left[\frac{u_{5}w_{5} + v_{5}}{u_{5}^{3}} \left(e^{-u_{5}t_{1}} + u_{5}t_{1} - 1 \right) - \frac{v_{5}}{2u_{5}} t_{1}^{2} - \frac{b_{1}''(\gamma)}{2\theta_{1}''(\gamma)} (T^{2} - t_{1}^{2}) + \frac{b_{1}''(\gamma) - \theta_{1}''(\gamma)(b_{1}''(\gamma)T + a_{1}''(\gamma))}{(\theta_{1}''(\gamma))^{3}} \left\{ 1 - e^{\theta_{1}''(\gamma)(T - t_{1})} \right\} \right]$$

$$\begin{aligned} HC_2''(\gamma) &= c_h \bigg[\frac{u_6 w_6 + v_6}{u_6^3} \big(e^{-u_6 t_1} + u_6 t_1 - 1 \big) - \frac{v_6}{2u_6} t_1^2 - \frac{b_2''(\gamma)}{2\theta_2''(\gamma)} (T^2 - t_1^2) \\ &+ \frac{b_2''(\gamma) - \theta_2''(\gamma) (b_2''(\gamma)T + a_2''(\gamma))}{(\theta_2''(\gamma))^3} \big\{ 1 - e^{\theta_2''(\gamma)(T - t_1)} \big\} \bigg] \end{aligned}$$

Therefore, the production cost $\tilde{PC} = \langle [PC_1(\alpha), PC_2(\alpha)], [PC'_1(\beta), PC'_2(\beta)], [PC''_1(\gamma), PC''_2(\gamma)] \rangle$ during the entire circle is given by

$$PC_{1}(\alpha) = c_{p} \left[\int_{0}^{t_{1}} \left\{ (m + la_{1}(\alpha)) - nq_{1}(t;\alpha) + lb_{1}(\alpha)t \right\} dt \right]$$

$$= c_{p} \left[(m + la_{1}(\alpha))t_{1} + lb_{1}(\alpha)\frac{t_{1}^{2}}{2} - n \int_{0}^{t_{1}} q_{1}(t;\alpha)dt \right]$$

$$= c_{p} \left[(m + la_{1}(\alpha))t_{1} + lb_{1}(\alpha)\frac{t_{1}^{2}}{2} - n \left\{ \frac{u_{1}w_{1} + v_{1}}{u_{1}^{3}} \left(e^{-u_{1}t_{1}} + u_{1}t_{1} - 1 \right) - \frac{v_{1}}{2u_{1}}t_{1}^{2} \right\} \right]$$

$$PC_{2}(\alpha) = c_{p} \left[\int_{0}^{t_{1}} \left\{ (m + la_{2}(\alpha)) - nq_{2}(t;\alpha) + lb_{2}(\alpha)t \right\} dt \right]$$

$$= c_{p} \left[(m + la_{2}(\alpha))t_{1} + lb_{2}(\alpha)\frac{t_{1}^{2}}{2} - n \int_{0}^{t_{1}} q_{2}(t;\alpha)dt \right]$$

$$= c_{p} \left[(m + la_{2}(\alpha))t_{1} + lb_{2}(\alpha)\frac{t_{1}^{2}}{2} - n \left\{ \frac{u_{2}w_{2} + v_{2}}{u_{2}^{3}} \left(e^{-u_{2}t_{1}} + u_{2}t_{1} - 1 \right) - \frac{v_{2}}{2u_{2}}t_{1}^{2} \right\} \right]$$

$$PC_{1}'(\beta) = c_{p} \left[(m + la_{1}'(\beta))t_{1} + lb_{1}'(\beta)\frac{t_{1}^{2}}{2} - n \left\{ \frac{u_{3}w_{3} + v_{3}}{u_{3}^{3}} \left(e^{-u_{3}t_{1}} + u_{3}t_{1} - 1 \right) - \frac{v_{3}}{2u_{3}}t_{1}^{2} \right\} \right]$$

$$PC_2''(\beta) = c_p \left[(m + la_2'(\beta))t_1 + lb_2'(\beta)\frac{t_1^2}{2} - n \left\{ \frac{u_4w_4 + v_4}{u_4^3} \left(e^{-u_4t_1} + u_4t_1 - 1 \right) - \frac{v_4}{2u_4}t_1^2 \right\} \right]$$

$$PC_1''(\gamma) = c_p \left[(m + la_1''(\gamma))t_1 + lb_1''(\gamma)\frac{t_1^2}{2} - n \left\{ \frac{u_5w_5 + v_5}{u_5^3} \left(e^{-u_5t_1} + u_5t_1 - 1 \right) - \frac{v_5}{2u_5}t_1^2 \right\} \right]$$

$$PC_2''(\gamma) = c_p \left[(m + la_2''(\gamma))t_1 + lb_2''(\gamma)\frac{t_1^2}{2} - n \left\{ \frac{u_6w_6 + v_6}{u_6^3} \left(e^{-u_6t_1} + u_6t_1 - 1 \right) - \frac{v_6}{2u_6}t_1^2 \right\} \right]$$

Therefore, the total average cost of the system during the entire circle can be obtained in the parametric form as $[\tilde{\Psi}]_{\alpha,\beta,\gamma} = \langle [\Psi_1(\alpha), \Psi_2(\alpha)], [\Psi_1'(\beta), \Psi_2'(\beta)], [\Psi_1''(\gamma), \Psi_2''(\gamma)] \rangle$, where $\Psi_1(\alpha) = \frac{HC_1(\alpha) + PC_1(\alpha) + C_0}{T}$, $\Psi_2(\alpha) = \frac{HC_2(\alpha) + PC_2(\alpha) + C_0}{T}$, $\Psi_1'(\beta) = \frac{HC_1'(\beta) + PC_1'(\beta) + C_0}{T}$, $\Psi_2'(\beta) = \frac{HC_2'(\beta) + PC_2'(\beta) + C_0}{T}$, $\Psi_1''(\gamma) = \frac{HC_1''(\gamma) + PC_1''(\gamma) + C_0}{T}$ and $\Psi_2''(\gamma) = \frac{HC_2''(\gamma) + PC_2''(\gamma) + C_0}{T}$.

Therefore, mathematically, the minimization problem concerning the production inventory model in the case of type-2 neutrosophic differentiability of $\tilde{q}(t)$ is obtained as follows:

$$\begin{cases}
Minimize \Psi_{1}(\alpha) \\
Minimize \Psi_{2}(\alpha) \\
Minimize \Psi_{1}'(\beta) \\
Minimize \Psi_{2}'(\beta) \\
Minimize \Psi_{1}''(\gamma) \\
Minimize \Psi_{2}''(\gamma) \\
Subject to T \ge t_{1} \ge 0 \\
0 \le \alpha, \beta, \gamma \le 1 \text{ with } \alpha + \beta + \gamma \le 3
\end{cases}$$
(50)

7. Used new De-Neutrosophication Method

To understand the neutrosophic outcome and to compare the results obtained in terms of neutrosophic numbers, it urges to assign a crisp value to the neutrosophic numbers in an appropriate means. The deneutrosophication technique then comes into the picture. The removal area method [12] is one of the popular methods. In this method, the de-neutrosophication value of a triangular single-valued neutrosophic number $\tilde{P}_{DN} = \langle (u, v, w : \epsilon), (x, y, z : \delta), (h, i, j : \gamma) \rangle$ is

$$D_{New}(\tilde{P}_{DN}, 0) = \frac{u + 2v + w + x + 2y + z + h + 2i + j}{12}$$
(51)

In the current article, the following de-neutrosophication technique is introduced:

Suppose $\tilde{A}_{TN} = (a_1, a_2, a_3; b_1, b_2, b_3; c_1, c_2, c_3)$ be a triangular neutrosophic number whose parametric form can be described by $\langle [A_1(\alpha), A_2(\alpha)], [A'_1(\beta), A'_2(\beta)], [A''_1(\gamma), A''_2(\gamma)] \rangle$, where $A_1(\alpha) = a_1 + \alpha(a_2 - a_1), A_2(\alpha) = a_3 - \alpha(a_3 - a_2), A'_1(\beta) = b_2 - \beta(b_2 - b_1), A'_2(\beta) = b_2 + \beta(b_3 - b_2), A''_1(\gamma) = c_2 - \gamma(c_2 - c_1)$ and $A''_2(\gamma) = c_2 + \gamma(c_3 - c_2)$. Then, the de-neutrosophication value of \tilde{A}_{TN} is denoted by A_{De-Neu} and is given by

$$A_{De-Neu} = \frac{A_1 + A_2 + A_3}{3} \tag{52}$$

where $A_1 = \alpha A_1(\alpha) + (1 - \alpha)A_2(\alpha)$, $A_2 = \beta A'_1(\beta) + (1 - \beta)A'_2(\beta)$ and $A_3 = \gamma A''_1(\gamma) + (1 - \gamma)A''_2(\gamma)$.

Using the proposed de-neutrosophication technique, the multi-objective optimization problem (30) for case 1 is transformed into a single objective crisp problem.

$$\begin{aligned}
\text{Minimize } \tilde{\Phi}_{De-New} \\
\text{where } \tilde{\Phi}_{De-New} &= \frac{\Phi_{11} + \Phi_{12} + \Phi_{13}}{3} \\
\Phi_{11} &= \alpha \Phi_1(\alpha) + (1 - \alpha) \Phi_2(\alpha) \\
\Phi_{12} &= \beta \Phi_1'(\beta) + (1 - \beta) \Phi_2'(\beta) \\
\Phi_{13} &= \gamma \Phi_1''(\gamma) + (1 - \gamma) \Phi_2''(\gamma) \\
\text{Subject to } T > t_1 > 0, \ 0 \le \alpha, \ \beta, \ \gamma \le 1.
\end{aligned}$$
(53)

For case 2, the multi-objective optimization problem (50) is transformed into the following single objective crisp problem

$$\begin{aligned} Minimize \ \tilde{\Psi}_{De-New} \\ where \ \tilde{\Psi}_{De-New} &= \frac{\Psi_{11}+\Psi_{12}+\Psi_{13}}{3} \\ \Psi_{11} &= \alpha \Psi_1(\alpha) + (1-\alpha)\Psi_2(\alpha) \\ \Psi_{12} &= \beta \Psi_1'(\beta) + (1-\beta)\Psi_2'(\beta) \\ \Psi_{13} &= \gamma \Psi_1''(\gamma) + (1-\gamma)\Psi_2''(\gamma) \\ Subject \ to \ T > t_1 > 0, \ 0 \le \alpha, \ \beta, \ \gamma \le 1. \end{aligned}$$

$$(54)$$

8. Numerical Simulation

8.1. Algorithm of the Numerical Solution

- Step 1 Input the numerical value of the crisp parameter $l, m, n, a, b, \theta, c_p, c_h$ and C_0 .
- Step 2 Solve the crisp minimization problem (7) and get the optimum average cost TAC, production cycle time t_1 and inventory cycle length T.
- Step 3 Take the demand-control parameters a and b and deterioration rate θ as the triangular neutrosophic number and consider the model in the neutrosophic arena. Go to step 4 or go to step 6.
- Step 4 Find the de-neutrosophication values of the neutrosophic parameters using the formula given in (51).
- Step 5 Solve the model with the de-neutrosophication values of the neutrosophic parameters along with other crisp inputs (called the old method).
- Step 6 Solve two minimization problems (53) and (54) (namely, Case 1 and Case 2) corresponding to two types of the generalized neutrosophic differentiability of the neutrosophic valued function $\tilde{q}(t)$.
- Step 7 Compare the results of Case 1, Case 2, and the old method for neutrosophic problems with the crisp method and get the optimum average cost.

Step 8 End.

The above-mentioned algorithm can be depicted visually in the following flow chart given in Figure 2.



8.2. Numerical results and graphical representation

In this subsection, four different problems are set for the numerical manipulation. Besides the crisp problem, three problems are considered from the data with neutrosophic uncertainty. Two types of neutrosophic differentiability are associated with two problems, namely Case 1 and Case 2. The problems named Case 1 and Case 2 are described and analyzed through the neutrosophic differential equation, and the optimum values of the decision variables and objective function are obtained using the proposed de-neutrosophication technique discussed in Section 7. Also, another problem (named the old method) is considered: taking de-neutrosophication of the given neutrosophic parameters before going for crisp-valued calculus-inspired dynamics of the system. For numerical simulation, the following inputs are considered:

- 1. For the crisp model, we take $m = 90, n = 0.06, l = 0.3, a = 40, b = 0.5, \theta = 0.05, c_p = 6, c_h = 1, C_0 = 140.$
- 2. For the neutrosophic model, the neutrosophic number \tilde{a} , \tilde{b} and $\tilde{\theta}$ are taken as a single-valued triangular neutrosophic number with nine components as
 - $\tilde{a} = (35, 40, 45; 41, 45, 49; 31, 37, 43),$
 - $\tilde{b} = (0.45, 0.50, 0.55; 0.54, 0.55, 0.56; 0.3, 0.45, 0.6)$ and
 - $\tilde{\theta} = (0.045, 0.050, 0.055; 0.054, 0.055, 0.056; 0.03, 0.045, 0.06)$

and the value of other parameters is taken as the same as in the crisp model.

The optimum value of the average cost (TAC^*) and the decision variables, namely, the total time cycle (T) and production time (t_1) is represented by Table 3. A graphical counterpart of the obtained results is also displayed through the bar diagram given in Figure 3.

Table 3. Optimum results for four different methods for solving the proposed EPQ model in crisp and neutrosophic arena





From Table 3 and Figure 3, it is perceived that the cost minimization objective is better fulfilled while considering the neutrosophic phenomena with neutrosophic calculus-oriented discussion and proposed de-neutrosophication technique. The old method with the removal of the area de-neutrosophication technique before going for the crisp calculus-oriented approach seems to give the most likely outcome of the crisp model. Therefore, the proposed approach to dealing with the dynamic of the inventory is established through numerical outcomes. Among the two cases of the proposed technique, Case 1 seems to be more effective in minimizing the TAC. However, the inventory cycle phase is seen to be very high in Case 1, which is concerned with the feasibility of the system. Therefore, Case 2 is fitted to the most

desired approach in the cost minimization objective with a feasible measure of the productive time and total lot-sizing cycle.



Figure 4. Total average cost versus total inventory cycle time.

Figure 4 shows the graph of the TAC concerning the lot-sizing cycle length. The initial trend of the graph of the TAC is decreasing against the total time cycle, and reaching the lowest value of 681.1805 at time 3.718347 months, the graph again increases.



Figure 5. Interdependency of average cost, production time, and time cycle.

Figure 5 shows the three-dimensional inter-dependence among the average cost, total time cycle, and production cycle. The locally convex nature of the graph around (1.867337, 3.718347, 681.1805) is spotted clearly in the figure.

8.3. Sensitivity Analysis with respect to deterministic parameters

In this section, a sensitivity analysis is performed for both cases on the crisp parameter by changing a parameter on a range of -15% to +30% while other parameters kept their original values. The sensitivity of the optimal results against the crisp parameters is given in Table 4. A graphical counterpart of the tabular display is presented in Figures 6 and 7.

| Crisp | Change | | Case 1 | | | Case 2 | |
|-------------|--------|---------|---------|----------|---------|---------|---------|
| parameters | in (%) | t_1^* | T^* | TAC^* | t_1^* | T^* | TAC^* |
| | +15 | 4.656 | 10.3219 | 328.92 | - | 2.5342 | 677.14 |
| | +10 | - | 0.8628 | 301.39 | 0.3922 | 2.8986 | 676.38 |
| m = 00 | +5 | 4.809 | 10.3031 | 315.80 | 1.4687 | 3.7814 | 668.75 |
| m = 90 | -5 | - | 10.2149 | 301.3885 | 46.7081 | 46.7081 | 368.63 |
| | -10 | 5.0546 | 10.2147 | 295.72 | 45.1739 | 45.1739 | 335.13 |
| | -15 | 5.0546 | 10.2146 | 295.72 | 43.5424 | 43,5424 | 290.33 |
| | +15 | 4.9276 | 10.3212 | 307.01 | 3.4011 | 5.3092 | 652.18 |
| | +10 | 4.9150 | 10.3086 | 307.72 | 3.1843 | 5.1119 | 652.86 |
| n=0.06 | +5 | - | 0.8628 | 301.39 | 3.0116 | 4.9562 | 653.47 |
| n=0.00 | -5 | 4.8777 | 10.2712 | 309.89 | 2.7487 | 4.7221 | 654.52 |
| | -10 | 4.8654 | 10.2589 | 310.61 | 2.6457 | 4.6312 | 654.97 |
| | -15 | 4.8532 | 10.2466 | 311.34 | 2.5568 | 4.5532 | 655.38 |
| | +15 | 4.9549 | 10.4763 | 312.18 | 2.0761 | 4.2152 | 661.43 |
| | +10 | 4.9335 | 10.4124 | 311.17 | 2.3046 | 4.3892 | 659.17 |
| 1-0.3 | +5 | - | 0.8628 | 301.39 | 2.5644 | 4.5895 | 656.71 |
| l=0.3 | -5 | 4.8681 | 10.2187 | 308.14 | 3.2451 | 5.1328 | 651.09 |
| | -10 | - | 0.8628 | 301.38 | 3.7611 | 5.5671 | 647.85 |
| | -15 | 4.8237 | 10.088 | 306.12 | 44.522 | 44.5220 | 383.48 |
| $c_p = 6$ | +15 | - | 0.8628 | 301.38 | - | 2.5342 | 677.14 |
| | +10 | 4.8900 | 10.2836 | 336.35 | - | - | - |
| | +5 | 4.8901 | 10.2836 | 322.75 | 1.7362 | 4.0398 | 668.66 |
| | -5 | 4.8901 | 10.2836 | 295.56 | 3.6488 | 5.2027 | 636.25 |
| | -10 | | 0.8628 | 301.39 | 4.0748 | 5.1733 | 616.31 |
| | -15 | 4.8902 | 10.2836 | 268.36 | 4.1871 | 4.7713 | 594.25 |
| | +15 | | 0.8628 | 322.26 | 3.7992 | 4.5065 | 689.20 |
| | +10 | 4.8901 | 10.2835 | 311.52 | 3.7494 | 4.8810 | 679.32 |
| $c_{1} = 1$ | +5 | 4.8901 | 10.2836 | 310.34 | 3.4733 | 5.0214 | 667.64 |
| $c_h = 1$ | -5 | - | 0.8628 | 294.43 | 1.8214 | 4.1753 | 637.55 |
| | -10 | - | 0.8628 | 287.48 | 0.1646 | 2.8207 | 614.75 |
| | -15 | - | 0.8628 | 280.52 | - | 2.7350 | 583.54 |
| | +15 | - | 0.8628 | 325.73 | 48.1973 | 48.1974 | 388.93 |
| | +10 | 4.8901 | 10.2836 | 310.52 | 3.1793 | 5.1926 | 656.82 |
| $C_0 = 140$ | +5 | - | 0.8628 | 309.50 | 3.0225 | 5.0097 | 655.45 |
| $C_0 = 140$ | -5 | 4.8901 | 10.2836 | 308.47 | 2.7183 | 4.6493 | 652.54 |
| | -10 | 4.8901 | 10.2836 | 307.79 | 2.5695 | 4.4703 | 651.01 |
| | -15 | 4.8901 | 10.2836 | 307.11 | 2.4222 | 4.2911 | 649.41 |

Table 4. Sensitivity of the optimum results concerning the crisp input

From Table 4, Figure 6, and Figure 7, the following points can be summarized:

(i) In the discussion, m denotes the fixed part of the production rate. It is seen that the average profit in the first case does not follow any monotonicity characteristic when the values of m increases.





Figure 6. Sensitivity analysis of case 1 of the crisp parameter on total average profit.

Figure 7. Sensitivity analysis of case 2 of the crisp parameter on total average profit.

Because the optimal solution of variant values of m corresponds to some distinguished production and decision tenures. The second case shows a specific pattern in this regard. It is noted that the average cost can be minimized through the minimization of the production rate.

- (ii) In the discussion, n denotes the coefficient of stock dependency of the production rate. As the value of n increases, the production rate should be lowered. However, in the first case, we find some infeasible cases regarding the pattern of optimal values of production and decision cycle to the variance of n. Thus, the impact of n on the cost, minimization is not well perceived in the first case. It is noted that the average cost can be lowered with the enhancement of the values of n in the second case.
- (iii) In the discussion, *l* denotes the coefficient of the demand dependency of the production rate. The model was built with the general perspective that the production rate should be increased as demand increases. Now, in the first case of the above discussion, there is no such straightforward conclusion on the impact of the demand on the minimization of the average profit. A significant result corre-

sponds to the second case of the above discussion. It is seen that the average cost increases with the enhancement of the demand rate.

(iv) The remaining crisp parameters in Table 4 are the production, holding, and setup costs of the items. Therefore, the average cost increases when the individual cost increases.

8.4. Sensitivity Analysis with respect to imprecise parameters

In this section, stability of the best obtained cost with optimal production and decision cycle length is examined in the context of imprecise parameters. The results regarding sensitivity of optimal solution with respect to neutrosophic valued parameters are displayed in Table 5. A graphical counterpart of the tabular display is presented in Figures 8 and 9.

| Imprecise | Change | Case 1 | | | | Case 2 | | |
|-----------------|--------|----------|-----------|---------|-----------|----------|---------|--|
| parameters | in (%) | t_1^* | T^* | TAC^* | t_1^* | T* | TAC^* | |
| | +15 | 5.236302 | 10.45742 | 331.31 | 2.871687 | 4.923177 | 658.26 | |
| | +10 | 0.000000 | 0.8412141 | 332.67 | 2.871785 | 4.892951 | 656.85 | |
| ã | +5 | 5.014098 | 10.35621 | 316.55 | 2.870835 | 4.861444 | 655.44 | |
| | -5 | 0.000000 | 0.8723352 | 286.77 | 2.866525 | 4.795275 | 652.59 | |
| | -10 | 0.000000 | 0.8830415 | 272.92 | 2.863420 | 4.760849 | 651.15 | |
| | -15 | 4.8901 | 10.2836 | 307.11 | 2.4222 | 4.2911 | 649.41 | |
| | +15 | 7.617859 | 15.06751 | 316.54 | 1.869831 | 4.148519 | 638.04 | |
| | +10 | 0.000000 | 0.8792371 | 286.96 | 2.267089 | 4.445040 | 643.72 | |
| ĩ | +5 | 4.842598 | 10.17940 | 309.85 | 2.598877 | 4.670114 | 649.01 | |
| ${	ilde b}$ | -5 | 0.000000 | 0.8430415 | 277.92 | 3.076379 | 4.925095 | 658.62 | |
| | -10 | 0.000000 | 0.873415 | 281.92 | 3.250088 | 4.974487 | 663.40 | |
| | -15 | 4.60248 | 10.19637 | 312.81 | 3.375978 | 4.976268 | 667.82 | |
| $\tilde{	heta}$ | +15 | 4.330256 | 9.149906 | 308.22 | 0.5962931 | 3.067986 | 624.63 | |
| | +10 | 4.502700 | 9.500159 | 308.43 | 1.291782 | 3.652540 | 636.28 | |
| | +5 | 4.688800 | 9.877092 | 308.74 | 2.036938 | 4.226846 | 646.03 | |
| | -5 | 5.108348 | 10.72296 | 309.70 | 3.871620 | 5.538776 | 660.22 | |
| | -10 | 5.345510 | 11.19878 | 310.41 | 45.49685 | 45.49685 | 415.23 | |
| | -15 | 5.603750 | 11.71507 | 311.31 | 44.12879 | 44.12879 | 421.71 | |

Table 5. Sensitivity of the optimum results concerning the Neutrosophic input

In the proposed model, three parameters have been taken imprecise. The demand is the most significant but volatile issue in an economic production quantity model. Therefore, both the demand impacting parameters have been taken triangular neutrosophic numbers for tackling impreciseness. Between these parameters, \tilde{a} denotes the imprecise demand potential, and \tilde{b} represents the coefficient of uncertain reliance of demand on the most crucial independent variable time. Also, we have considered the rate of deterioration as imprecise parameters $\tilde{\theta}$. The best results in an imprecise environment correspond to the imprecise inputs $\tilde{a} = (35, 40, 45; 41, 45, 49; 31, 37, 43)$, $\tilde{b} = (0.45, 0.50, 0.55; 0.54, 0.55, 0.56; 0.3, 0.45, 0.6)$ and $\tilde{\theta} = (0.045, 0.050, 0.055; 0.054, 0.055, 0.056; 0.03, 0.045, 0.06)$ in both cases of neutrosophic differentiation. The sensitivity analysis with respect to these parameters has been performed varying the impacting parameters between -15% and 15% of the mentioned values. The changing the values with given ranges, not only the value is scaled but also dispersion of data due imprecision is also scaled. The results in Table 5 can be interpreted as follows:







Figure 9. Sensitivity analysis of case 2 of the Neutrosophic parameter on total average profit.

- (i) In case of neutrosophic derivative of first type, we did not find any specific pattern of changes in average costs with respect to the demand potential \tilde{a} . Furthermore, the productive cycle becomes infeasible in many occasions connected to Case 1. However, in Case 2, a crystal pattern is obtained regarding the impact of the demand potential on average cost. The average cost can be made lowered by decreasing demand potential. In this case, the ambiguities associated with demand prediction increases with production potential resulting negative impression on cost reduction goal. Also, the production cycle and complete lot cycle are increased with the deviation of imprecise data.
- (ii) In this imprecise EPQ model, \tilde{b} represents the coefficient of variability of demand on time. In case of neutrosophic derivative of first type, we did not find any specific pattern of changes in average costs against variance of \tilde{b} . Furthermore, the productive cycle becomes infeasible in many occasions connected to Case 1. However, in Case 2, the results in Table 5 reflect reverse pattern compared to the pattern obtained for \tilde{a} . The cost can be minimized by increasing values of \tilde{b} . The observation can be interpreted that increasing time influence of the demand rate favors the cost reduction goal.

The production cycle and whole lot cycle also follow the pattern for average cost.

(iii) The deterioration of products during inventory carrying period is a natural phenomenon. In this paper, deterioration rate is described by an uncertain parameter $\tilde{\theta}$. In both cases of the neutrosophic derivative of inventory function of time, average cost, production and decision lot sizes decrease with the increasing nature of $\tilde{\theta}$ and associated impressions.

9. Major research findings and managerial intuitions

In the first part of this section, we summarize all the research outcomes from the proposed model. The managerial implications corresponding to the outcomes are listed subsequently. Fundamental observations in numerical simulation can be briefed as follows:

- (i) Average cost increases robustly accordingly the primary part of the production rate in all the discussed models. However, in both cases of imprecise decision environment, optimal manufacturing cycle and overall decision cycle become infeasible with respect to many values of m. Thus, the obtained solutions are locally optimal and highly sensitive with respect to the demand potential.
- (ii) In proposed model, n denotes the coefficient of stock controlling the production rate. The numerical results show that cost can be minimized by increasing the influence of stock on production process.
- (iii) The results regarding l shows that average cost increases as the control of demand on the production process increases.
- (iv) The imprecise decision-making phenomena with triangular neutrosophic number reduce the cost effectively compared two decision making environment. Among mentioned approaches of neutrosophic decision making phenomena, the neutrosophic differential equation of type 2 is established smarter.
- (v) The average cost can be made lowered by decreasing demand potential. However, increasing time influence of the demand rate favors the cost reduction goal.

At the end of this section the managerial implications are decoded as follows:

- (i) The decision maker must install cohesion between on hand stock and production process. The average cost can be made significantly reduced by a smart managerial strategy which considers strong influence of the inventory on manufacturing rate.
- (ii) The demand is imprecise and indeterministic in nature. As a consequence, much impacts of demand on production process includes additional cost. Therefore, the smart decision maker must set policy such that the production process can be impacted a very little with the imprecisions involved with demand pattern.
- (iii) Also, the decision maker can reduce the average cost by increasing the weight of influence of time on the demand pattern.

10. Conclusions

In this paper, we use an EPQ model with a time-dependent demand rate, a demand-, and stock-dependent production rate, and a constant rate of deterioration. The hypothesis was built on a very general perception. In the newly organized manufacturing-supply sector, demand increases as time goes by. Also, the production rate varies according to the demand rate and the stock already in the warehouse. During the inventory-carrying process, the produced items deteriorate gradually as a natural cause. Our objective was to address the impact of the mentioned issues on cost reduction. Furthermore, an additional query was about the involvement of uncertainty regarding the decision environment. Therefore, in this paper, the proposed model has been formulated mathematically in a crisp environment. With the theory of differential equations and calculus, the criteria for the optimality of the objective function have been discussed in a crisp environment. Then, the uncertainty regarding real-world production-inventory management has been tackled using neutrosophic numbers and the theory of the neutrosophic differential equation. In the end, a de-neutrosophication formula has been used for comparing results in distinct cases of uncertain phenomena as well as their crisp counterparts. The numerical results provide some of the finest insights regarding the proposed model. First, the demand enhancement is not favorable for the cost minimization objective. Another significant observation is that stock in the showroom can favor the cost minimization objective. Furthermore, the proposed approach of the neutrosophic differential equation as a tool to describe the EPQ model and the proposed de-neutrosophication formula provide better results compared to the crisp and existing methods to deal with an EPQ model under uncertainty.

At the end of the discussion, it is to be mentioned that the theory of neutrosophic numbers has been limited in some multi-criteria decision-making problems in the recently introduced literature. However, it has a wider scope to describe the uncertainty regarding the dynamics of problems in operational research, especially in inventory theory. In the future, the notions of neutrosophic calculus and neutrosophic differential equations must be improved; therefore, a non-linear inventory model with more realistic assumptions can be fitted to be analyzed through it. Furthermore, the proposed model can be modified with variable rates of deterioration and preservation measures, etc., for more reliable mathematical models in the future.

Acknowledgement

The authors are grateful to two anonymous reviewers for their valuable comments and suggestions made on the previous draft of this manuscript.

References

- ABU ARQUB, O. Adaptation of reproducing kernel algorithm for solving fuzzy fredholm-volterra integrodifferential equations. *Neural Computing and Applications* 28 (2017), 1591–1610.
- [2] ABU ARQUB, O., SINGH, J., AND ALHODALY, M. Adaptation of kernel functions-based approach with atangana-baleanucaputo distributed order derivative for solutions of fuzzy fractional volterra and fredholm integrodifferential equations. *Mathematical Methods in the Applied Sciences* 46, 7 (2023), 7807–7834.
- [3] ABU ARQUB, O., SINGH, J., MAAYAH, B., AND ALHODALY, M. Reproducing kernel approach for numerical solutions of fuzzy fractional initial value problems under the mittag–leffler kernel differential operator. *Mathematical Methods in the Applied Sciences* 46, 7 (2023), 7965–7986.
- [4] AGBOOLA, A., AND AKINLEYE, S. Neutrosophic vector spaces. Neutrosophic Sets and Systems 4 (2014), 9-18.

- [5] ALI, I., GUPTA, S., AND AHMED, A. Multi-objective linear fractional inventory problem under intuitionistic fuzzy environment. International Journal of System Assurance Engineering and Management 10 (2019), 173–189.
- [6] ALI, M., AND SMARANDACHE, F. Complex neutrosophic set. Neural computing and applications 28 (2017), 1817–1834.
- [7] ALSHAMMARI, M., AL-SMADI, M., ARQUB, O. A., HASHIM, I., AND ALIAS, M. A. Residual series representation algorithm for solving fuzzy duffing oscillator equations. *Symmetry* 12, 4 (2020), 572.
- [8] BAG, S., CHAKRABORTY, D., AND ROY, A. A production inventory model with fuzzy random demand and with flexibility and reliability considerations. *Computers & Industrial Engineering 56*, 1 (2009), 411–416.
- [9] BARMAN, H., PERVIN, M., ROY, S. K., AND WEBER, G.-W. Back-ordered inventory model with inflation in a cloudy-fuzzy environment. *Journal of Industrial & Management Optimization* 17, 4 (2021), 1913–1941.
- [10] CÁRDENAS-BARRÓN, L. E. Economic production quantity with rework process at a single-stage manufacturing system with planned backorders. *Computers & Industrial Engineering* 57, 3 (2009), 1105–1113.
- [11] CHAKRABORTY, A., ET AL. A new score function of pentagonal neutrosophic number and its application in networking problem. *International Journal of Neutrosophic Science 1*, 1 (2020), 40–51.
- [12] CHAKRABORTY, A., MONDAL, S. P., AHMADIAN, A., SENU, N., ALAM, S., AND SALAHSHOUR, S. Different forms of triangular neutrosophic numbers, de-neutrosophication techniques, and their applications. *Symmetry 10*, 8 (2018), 327.
- [13] CUNHA, L. R. A., DELFINO, A. P. S., DOS REIS, K. A., AND LEIRAS, A. Economic production quantity (epq) model with partial backordering and a discount for imperfect quality batches. *International Journal of Production Research 56*, 18 (2018), 6279–6293.
- [14] DE, S. K., AND SANA, S. S. Fuzzy order quantity inventory model with fuzzy shortage quantity and fuzzy promotional index. *Economic Modelling 31* (2013), 351–358.
- [15] DE, S. K., AND SANA, S. S. A multi-periods production-inventory model with capacity constraints for multi-manufacturers-a global optimality in intuitionistic fuzzy environment. *Applied Mathematics and Computation* 242 (2014), 825–841.
- [16] DE, S. K., AND SANA, S. S. Backlogging eoq model for promotional effort and selling price sensitive demand-an intuitionistic fuzzy approach. Annals of Operations Research 233 (2015), 57–76.
- [17] DEBNATH, B., MAJUMDER, P., AND BERA, U. Multi-objective sustainable fuzzy economic production quantity (sfepq) model with demand as type-2 fuzzy number: A fuzzy differential equation approach. *Hacettepe Journal of Mathematics and Statistics* 48, 1 (2019), 112–139.
- [18] GARAI, T., CHAKRABORTY, D., AND ROY, T. K. A multi-item generalized intuitionistic fuzzy inventory model with inventory level dependent demand using possibility mean, variance and covariance. *Journal of Intelligent & Fuzzy Systems* 35, 1 (2018), 1021– 1036.
- [19] GIRI, S. K., GARAI, T., GARG, H., AND ISLAM, S. Possibilistic mean of generalized non-linear intuitionistic fuzzy number to solve a price and quality dependent demand multi-item inventory model. *Computational and applied Mathematics* 40, 4 (2021), 110.
- [20] HAQUE, R., PERVIN, M., AND MONDAL, S. P. A sustainable manufacturing-remanufacturing inventory model with price and green sensitive demand for defective and usable items. *RAIRO-Operations Research* 58, 4 (2024), 3439–3467.
- [21] HOJATI, M. Bridging the gap between probabilistic and fuzzy-parameter eoq models. *International Journal of Production Economics* 91, 3 (2004), 215–221.
- [22] HSU, L.-F., AND HSU, J.-T. Economic production quantity (epq) models under an imperfect production process with shortages backordered. *International Journal of Systems Science* 47, 4 (2016), 852–867.
- [23] KARMAKAR, S., DE, S. K., AND GOSWAMI, A. A pollution sensitive remanufacturing model with waste items: triangular dense fuzzy lock set approach. *Journal of Cleaner Production 187* (2018), 789–803.
- [24] KESHAVARZFARD, R., MAKUI, A., TAVAKKOLI-MOGHADDAM, R., AND TALEIZADEH, A. A. Optimization of imperfect economic manufacturing models with a power demand rate dependent production rate. *Sādhanā* 44 (2019), 1–19.
- [25] KHAN, M., HUSSAIN, M., AND CÁRDENAS-BARRÓN, L. E. Learning and screening errors in an epq inventory model for supply chains with stochastic lead time demands. *International Journal of Production Research* 55, 16 (2017), 4816–4832.
- [26] MAHATA, G. C., DE, S. K., BHATTACHARYA, K., AND MAITY, S. Three-echelon supply chain model in an imperfect production system with inspection error, learning effect, and return policy under fuzzy environment. *International Journal of Systems Science: Operations & Logistics 10*, 1 (2023), 1962427.
- [27] MAITI, A. K. Cloudy fuzzy inventory model under imperfect production process with demand dependent production rate. *Journal of Management Analytics* 8, 4 (2021), 741–763.
- [28] MAJUMDER, P., BERA, U., AND MAITI, M. An epq model of deteriorating items under partial trade credit financing and demand declining market in crisp and fuzzy environment. *Procedia Computer Science* 45 (2015), 780–789.
- [29] MANNA, A. K., RAHMAN, M. S., SHAIKH, A. A., BHUNIA, A. K., AND KONSTANTARAS, I. Modeling of a carbon emitted production inventory system with interval uncertainty via meta-heuristic algorithms. *Applied Mathematical Modelling 106* (2022), 343–368.
- [30] MARCHI, B., ZANONI, S., AND JABER, M. Y. Economic production quantity model with learning in production, quality, reliability and energy efficiency. *Computers & Industrial Engineering 129* (2019), 502–511.
- [31] MOI, S., BISWAS, S., AND PAL, S. Neutrosophic linear differential equation with a new concept of neutrosophic derivative. In *Neutrosophic Operational Research: Methods and Applications*. Springer, 2021, pp. 395–410.
- [32] MOI, S., BISWAS, S., AND PAL, S. Second-order neutrosophic boundary-value problem. Complex & Intelligent Systems 7 (2021), 1079–1098.
- [33] MOMENA, A. F., HAQUE, R., RAHAMAN, M., SALAHSHOUR, S., AND MONDAL, S. P. The existence and uniqueness conditions for solving neutrosophic differential equations and its consequence on optimal order quantity strategy. *Logistics* 8, 1 (2024),

18.

- [34] MOMENA, A. F., RAHAMAN, M., HAQUE, R., ALAM, S., AND MONDAL, S. P. A learning-based optimal decision scenario for an inventory problem under a price discount policy. *Systems* 11, 5 (2023), 235.
- [35] MONDAL, B., GARAI, A., MUKHOPADHYAY, A., AND MAJUMDER, S. K. Inventory policies for seasonal items with logisticgrowth demand rate under fully permissible delay in payment: a neutrosophic optimization approach. *Soft Computing* 25, 5 (2021), 3725–3750.
- [36] MULLAI, M., AND SURYA, R. Neutrosophic eoq model with price break. Neutrosophic sets and systems 19, 1 (2018), 24–28.
- [37] MULLAI, M., AND SURYA, R. Neutrosophic inventory backorder problem using triangular neutrosophic numbers. *Neutrosophic Sets and Systems 31*, 1 (2020), 148–155.
- [38] NOBIL, A. H., NOBIL, E., SEDIGH, A. H. A., CÁRDENAS-BARRÓN, L. E., GARZA-NÚÑEZ, D., TREVIÑO-GARZA, G., CÉSPEDES-MOTA, A., DE JESÚS LOERA-HERNÁNDEZ, I., AND SMITH, N. R. Economic production quantity models for an imperfect manufacturing system with strict inspection. *Ain Shams Engineering Journal 15*, 5 (2024), 102714.
- [39] PAL, S., CHAKRABORTY, A., ET AL. Triangular neutrosophic-based eoq model for non-instantaneous deteriorating item under shortages. Am J Bus Oper Res 1, 1 (2020), 28–35.
- [40] PARK, K. S. Fuzzy-set theoretic interpretation of economic order quantity. *IEEE Transactions on systems, Man, and Cybernetics* 17, 6 (1987), 1082–1084.
- [41] PURI, M. L., AND RALESCU, D. A. Differentials of fuzzy functions. *Journal of Mathematical Analysis and Applications 91*, 2 (1983), 552–558.
- [42] RAHAMAN, M., ALAM, S., HAQUE, R., SHAIKH, A. A., BEHERA, P. K., AND MONDAL, S. P. An economic production quantity model for an imperfect production system with selling price, advertisement frequency and green-level dependent demand. *Electronic Commerce Research* (2024), 1–28.
- [43] RAHAMAN, M., HAQUE, R., ALAM, S., ZUPOK, S., SALAHSHOUR, S., AZIZZADEH, F., AND MONDAL, S. P. Solvability criteria for uncertain differential equations and their applicability in an economic lot-size model with a type-2 interval phenomenon. *Symmetry 15*, 10 (2023), 1883.
- [44] RAHAMAN, M., MONDAL, S. P., ALAM, S., AND DE, S. K. A study of a lock fuzzy epq model with deterioration and stock and unit selling price-dependent demand using preservation technology. *Soft Computing* 26, 6 (2022), 2721–2740.
- [45] SADEGHI, J., MOUSAVI, S. M., AND NIAKI, S. T. A. Optimizing an inventory model with fuzzy demand, backordering, and discount using a hybrid imperialist competitive algorithm. *Applied Mathematical Modelling* 40, 15-16 (2016), 7318–7335.
- [46] SAHOO, S., ACHARYA, M., AND PATNAIK, S. Sustainable intuitionistic fuzzy inventory models with preservation technology investment and shortages. *International Journal of Reasoning-based Intelligent Systems* 14, 1 (2022), 8–18.
- [47] SALAMA, A., AND ALBLOWI, S. Neutrosophic set and neutrosophic topological spaces. *IOSR Journal of Mathematics 3*, 4 (2012), 31 35.
- [48] SARKAR, B., AND MAHAPATRA, A. S. Periodic review fuzzy inventory model with variable lead time and fuzzy demand. *International Transactions in Operational Research* 24, 5 (2017), 1197–1227.
- [49] SMARANDACHE, F. Neutrosophic precalculus and neutrosophic calculus: neutrosophic applications. Infinite Study, 2015.
- [50] SMARANDACHE, F., AND ALI, M. Neutrosophic triplet group. Neural Computing and Applications 29, 7 (2018), 595-601.
- [51] SON, N. T. K., DONG, N. P., LONG, H. V., KHASTAN, A., ET AL. Linear quadratic regulator problem governed by granular neutrosophic fractional differential equations. *ISA transactions* 97 (2020), 296–316.
- [52] SUMATHI, I., AND ANTONY CRISPIN SWEETY, C. New approach on differential equation via trapezoidal neutrosophic number. Complex & Intelligent Systems 5 (2019), 417–424.
- [53] SUMATHI, I., AND PRIYA, V. M. A new perspective on neutrosophic differential equation. Infinite Study, 2018.
- [54] SUPAKAR, P., MAHATO, S. K., AND PAL, P. Intuitionistic fuzzy inventory model with deterioration incorporating advance payment with time-dependent demand. *International Journal of Applied and Computational Mathematics* 7, 6 (2021), 228.
- [55] TALEIZADEH, A. A., NAGHAVI-ALHOSEINY, M.-S., CÁRDENAS-BARRÓN, L. E., AND AMJADIAN, A. Optimization of price, lot size and backordered level in an epq inventory model with rework process. *RAIRO-Operations Research* 58, 1 (2024), 803–819.
- [56] TALEIZADEH, A. A., SOLEYMANFAR, V. R., AND GOVINDAN, K. Sustainable economic production quantity models for inventory systems with shortage. *Journal of cleaner production 174* (2018), 1011–1020.
- [57] TALEIZADEH, A. A., WEE, H.-M., AND JALALI-NAINI, S. G. Economic production quantity model with repair failure and limited capacity. *Applied Mathematical Modelling* 37, 5 (2013), 2765–2774.
- [58] TENG, J.-T., LOU, K.-R., AND WANG, L. Optimal trade credit and lot size policies in economic production quantity models with learning curve production costs. *International Journal of Production Economics 155* (2014), 318–323.
- [59] WANG, H., SMARANDACHE, F., ZHANG, Y., AND SUNDERRAMAN, R. Single valued neutrosophic sets. Infinite study, 2010.
- [60] WIDYADANA, G. A., AND WEE, H. M. An economic production quantity model for deteriorating items with multiple production setups and rework. *International Journal of Production Economics* 138, 1 (2012), 62–67.
- [61] YE, J. Trapezoidal neutrosophic set and its application to multiple attribute decision-making. *Neural computing and Applications* 26 (2015), 1157–1166.